Marie Skłodowska Curie Action

# WINDMILL

Machine Learning for Wireless Communications

# H2020-MSCA-ITN-ETN Grant Agreement Number: 813999



# WP6–System-wide Cognitive Optimisation Schemes

D6.1–State of the art on context acquisition & anticipatory optimisation techniques for network optimisation

Contractual Delivery Date:	Month 20 (August 2020)
Actual Delivery Date:	August 31, 2020
Responsible Beneficiary:	UNIPD
Contributing Beneficiaries:	AALTO, BOSCH
Dissemination Level:	Public
Version:	Draft



### PROPRIETARY RIGHTS STATEMENT

This project has received funding from the European Union's Horizon 2020 research and innovation programme under the Marie Skłodowska-Curie grant agreement No 813999.





PROPRIETARY RIGHTS STATEMENT

This project has received funding from the European Union's Horizon 2020 research and innovation programme under the Marie Skłodowska-Curie grant agreement No 813999.



### **Document Information**

Document ID:	WP6/D6.1
Version Date:	August 31, 2020
Total Number of	61
Pages:	
Abstract:	Future wireless networks attempt to combine different types of communication for various applications thereby leading to a high number of devices needed to be integrated and handled effi- ciently. Hence, the need for network optimization in future wire- less networks, such as 5G networks and beyond, has become very important as large number of devices requesting a certain data rate would need high amount of bandwidth and advanced re- source management algorithms to maintain an appreciable qual- ity of service and quality of experience. In order to handle largely variable requirements, prediction and estimation of wireless net- work state parameters, such as channel state, user data require- ment, user mobility, are needed to calculate optimal network con- figurations for a certain application at certain time for a certain user. So, in this document, we describe the different method- ologies and schemes proposed for anticipatory or predictive net- working and optimization schemes proposed based on the differ- ent types of anticipatory information from the wireless networks.
Keywords:	Anticipatory Networking, Machine Learning, 5G Networks, Con- textual Information, Optimization Schemes

## **Authors**

Full name	Beneficiary/ Organisation	e-mail	Role
Anay Ajit Deshpande	UNIPD	anayajit.deshpande @unipd.it	Contributor
Salman Mohebi	UNIPD	salman.mohebiganjabadi @unipd.it	Contributor
Dariush Salami	AALTO	dariush.salami@aalto.fi	Contributor
Pedro Maia de Sant Ana	Bosch	Pedro.MaiadeSantAna @de.bosch.com	Contributor



# Reviewers

Full name	Beneficiary/ Organi- sation	e-mail	Date
Andrea Zanella	UNIPD	andrea.zanella @unipd.it	Supervisor



# **Table of Contents**

1	Intro	oduction	13
2	Moti	vation	14
3	Con 3.1 3.2 3.3	textual Information for Wireless NetworksGeographical Context	<b>15</b> 15 16 16 17 17 18 18 18 18
	3.4	Social ContextSocial Context based caching3.4.1Social Context based Mobility Prediction3.4.2Social Context based Mobility Prediction3.4.3Social Context based Access Control	19 19 20 20
4	<b>Anti</b> 4.1 4.2 4.3	cipatory TechniquesTime Series Prediction	<b>21</b> 22 22 24 24 25 27
5		eless Network Optimization Techniques using Anticipatory and Context Away         oniques         Optimization Approaches for Convex Optimization Problem         Optimization Approaches for Non-Convex Optimization Problem         5.2.1       Model Predictive Control         5.2.2       Reinforcement Learning         5.2.3       Optimization Approaches based on Data Driven Model Prediction         5.2.4       Optimization Approaches based on Game theory	re 30 31 31 32 35 35
6	<b>App</b> 6.1 6.2 6.3 6.4	Iications of Optimization Techniques in Wireless Networks         Physical Layer         MAC Layer         Network Layer         Application Layer	<b>37</b> 37 38 39 39



### 7 References



# List of Figures

4.1	Example of state flow diagram of a Markov Chain (MC).	23
4.2	Example of k-means algorithm on a synthetic dataset after six steps with k equals to 4	26
4.3	Examples of applying the Support Vector Machine (SVM) classifier with a linear and Gaussian kernels on a linearly non-separable synthetic dataset.	28
5.1	Action-reward loop in Reinforcement Learning (RL)	32



List of Acronyms and Abbreviations

- **AP** Access Point
- Aol Age of Information
- **ANN** Artificial Neural Network
- AGM Arithmetic-Geometric Mean
- AOA Angle of Arrival
- ALQ Average Link Quality
- ARIMA Auto Regression Integrated Moving Average
- AWGN Additive White Gaussian Noise
- **BS** Base Station
- **BER** Bit Error Rate
- **CF** Collaborative Filtering
- **CR** Cognitive Radio
- **CRN** Cognitive Radio Network
- **CSI** Channel State Information
- **CNN** Convolutional Neural Network
- **CoMP** Coordinated Multipoint
- **CSMA** Carrier Sense Multiple Access
- CL-IDA Cross-layer Interference and Delay Aware Metric
- **DDPG** Deep Deterministic Policy Gradient
- **DNN** Deep Neural Network
- **DP** Dynamic Programming
- **DQN** Deep Q-Network
- **DRL** Deep Reinforcement Learning
- **DSP** Digital Signal Processor
- DTMC Discrete Time Markov Chain
- D2D Device to Device
- **DBNG** Deep Belief Network and Gaussian Model
- Dissemination Level: Public.



- **ESN** Echo State Network
- ETX Expected Transmission Count
- **ETT** Expected Transmission Time
- FDD Frequency Division Duplexing
- FEV Fully Electric Vehicle
- FNN Feed Forward Neural Network
- FDD Frequency Division Duplexing
- **GAN** Generative Adversarial Network
- GMM Gaussian Mixed Model
- GPR Gaussian Process Regression
- GPU Graphics Processing Unit
- **GBSM** Geometry based Stochastic Model
- **GBDT** Gradient Boosting Decision Tree
- GECR Genetic algorithm-based energy-efficient clustering and routing algorithm
- **GPS** Global Positioning System
- HCRAN Heterogeneous Cloud Radio Access Network
- HetNet Heterogeneous Networks
- HGMM Hidden Gaussian Markov Model
- HMM Hidden Markov Model
- **IRS** Intelligent Reflecting Surface
- IoV Internet of Vehicles
- **IoT** Internet of Things
- KNN K-Nearest Neighbors
- **LSTM** Long Short-Term Memory
- LP Linear Program
- ILP Integer Linear Program
- MILP Mixed-Integers Linear Program
- MAC Medium Access Control
- Dissemination Level: Public.



MC Markov Chain **MCS** Modulation and Coding Scheme **MDP** Markov Decision Process **ML** Machine Learning **MISO** Multiple-input Single-output **MPC** Model Predictive Control **MF** Matrix Factorization **MIMO** Multiple-Input Multiple-Output mmWave Millimeter Wave **MIP** Mixed Integer Programming **MANET** Mobile Ad-hoc Network **MLR** Multinomial Logistic Regression MIMO Multiple Input Multiple Output **NLoS** None Line of Sight **NOMA** Non-Orthogonal Multiple Access **NB** Naive Bayes **OFDMA** Orthogonal Frequency Division Multiple Access **OLSR** Optimized Link State Routing **OS** Operating System QoS Quality of Service **QoE** Quality of Experience **RAN** Radio Access Network **RB** Resource Block **RF** Radio Frequency **RL** Reinforcement Learning **RSSI** Received Signal Strength Indicator **RRH** Radio Remote Head **RNN** Recurrent Neural Network



- **RRM** Radio Resource Management
- **RBF-NN** Radial Basis Function Neural Network
- RLRP Radio Link Reliability Prediction
- RSU Road Side Unit
- **SISO** Single-Input Single-Output
- SJE Stationary Joint Eigentensor
- SVD Singular Value Decomposition
- SBS Small Base Station
- SVM Support Vector Machine
- SVR Support Vector Regression
- SP Service Provider
- SCA Successive Convex Approximation
- SDS-TWR Symmetric Double Sided Two Way Ranging
- SMS Short Message Service
- **SDN** Software Defined Network
- SINR Signal-To-Interference-plus-Noise
- **TOA** Time Of Arrival
- **TDOA** Time Difference of Arrival
- TPU Tensor Processing Unit
- **MP-TCP** Multipath Transmission Control Protocol
- **UAV** Unmanned Aerial Vehicle
- **UE** User Equipments
- **URLLC** Ultra Reliable Low Latency Communication
- UWB Ultra Wide Band
- V2I Vehicle to Infrastructure
- V2V Vehicle to Vehicle
- V2X Vehicle-to-X
- VANET Vehicular Ad Hoc Network
- Dissemination Level: Public.



VM Virtual Machine
WLAN Wireless Local Area Network
WSAN Wireless Sensor and Actuator Network
WSN Wireless Sensor Network
WMN Wireless Mesh Network



# **1. Introduction**

Future wireless networks attempt to combine different types of communication for various applications thereby leading to a high number of devices needed to be integrated and handled efficiently. Hence, the need for network optimization in future wireless networks, such as 5G networks, has become very important as large number of devices requesting a certain data rate would need high amount of bandwidth and advanced resource management algorithms to maintain an appreciable quality of service and quality of experience. Satisfying these requirements with traditional optimization schemes would result in high computational costs to find optimal solutions for different types of applications. In order to handle largely variable requirements, prediction and estimation of wireless network state parameters, such as channel state, user data requirement, user mobility, are needed to calculate optimal network configurations for a certain application at certain time for a certain user. To predict the network behaviour, the system must continuously track the performance of the network.

Predictions of such network state parameters can be performed using multiple different techniques such as Time Series Predictive Modelling, Probabilistic Forecasting, Similarity Based Classification and Regression Analysis [1]. In Time Series Predictive Modelling, network parameters are estimated by modelling wireless networks as auto-regressive and moving average model [2]. Probabilistic Forecasting shows that parameters can be estimated by modelling networks as probability processes such as Markov Process [3]. Similarity Based Classification finds relations between parameters and gather similar parameters to predict future network parameters [4]. Lastly, Regression Analysis or Machine Learning learns about the parameters and determines dependencies so as to predict different network parameters [5–8].

Hence, in this document, we describe the different methodologies and schemes proposed for anticipatory networking and optimization schemes based on the different types of anticipatory information from the wireless networks. Chapter 2 mentions the motivation for anticipatory networking and optimization based on the anticipatory schemes. Chapter 3 describes the methodologies to gather different types of contextual information for anticipatory techniques. Chapter 4 describes the techniques to generate anticipatory information from the working context. Chapter 5 discusses the different types of techniques designed to optimize the system based on the anticipatory information. Chapter 6 discusses the various applications that are tackled by using the different proposed optimization schemes.



# 2. Motivation

Predictions of parameters in wireless networks is a huge challenge, due to the inherent randomness of these systems. To overcome such a challenge, knowledge regarding the wireless network parameters has to be gained. Also, such knowledge has to be interpreted by the system so as to create optimal configurations based on the requirements of the network. To do so, large amount of data is necessary to be analyzed and models have to be created based on such analysis. The state of the art systems such as Time Series Predictive Modelling and Probability Forecasting depend on accurate modelling of all network parameters but are unable to find relations between such parameters implicitly. Hence, they are tedious to pursue for a dynamic network such as the 5G system, which will incorporate multiple different applications. Also, similarity based classification is hindered as predictions of parameters is difficult for the model if there are no instances for such predictions in the data. Hence, it is limited to certain network optimization configurations and cannot account for unknown interaction between different parameters for various network configurations, which is needed in 5G networks when dealing with various devices for different applications. The most ideal approach for network optimization is the usage of Machine Learning especially Artificial Neural Networks or Deep Neural Networks, that have the capabilities to gather and learn about large amount of information of network parameters and create relations between such parameters to predict the optimal network configurations for the current state of the system. To learn and predict network parameters, artificial neural networks employ different methodologies such as supervised learning, unsupervised learning, reinforcement learning and transfer learning. These techniques are incorporated to generate models which are highly efficient for prediction of wireless network parameters as well as determining optimal network configurations for such parameters.

This approach is novel due to the fact that future networks need to integrate multiple different types of devices and need to analyze large amount of information from such devices which is not quite prominent in the current network setup. Also this approach is possible due to the advancement in processors, especially GPUs, which enables to process large amount of information using parallel computing. Hence, usage of deep neural networks to predict wireless network parameters for optimization is essential for future wireless networks such as 5G and beyond 5G systems.



## **3. Contextual Information for Wireless Networks**

Wireless Networks incorporate a ton of contextual information which can be very useful for their optimization. User information such as position and mobility as well as Received Signal Strength Indicator (RSSI) and Channel State Information (CSI) provide essential information which can be used by the wireless systems to allocate resources and improve performance of the wireless network.

In this chapter, different types of contextual information that can be extracted from the wireless network are discussed. The chapter also discusses the works that are proposed to gather these contextual information and the use of the information in different applications. Section 3.1 discusses the techniques to gather and predict geographical contextual information. Section 3.2 discusses the techniques to gather and predict wireless link related information. Section 3.3 discusses the methodologies to predict user traffic in the network. Section 3.4 discusses the methodologies to gather social information about the users and applications related to this social information.

### **3.1. Geographical Context**

Geographical context refers to the user information related to the physical environment. Prominently geographical context includes the position and mobility information regarding the users in the network, and also their trajectories. The difference between mobility and trajectory is that mobility refers to the prediction of the next position of the user, while trajectory refers to the prediction of the whole path of the mobile user. This information is essential while determining the configuration of the network and the requirements of the users so as to provide maximum Quality of Service (QoS) and Quality of Experience (QoE) to the users.

### 3.1.1. Position and Mobility Prediction

Position and mobility information is one of the most important contextual information to configure and manage a wireless system. Traditionally, the users periodically communicate their position information to the network infrastructure. But with the increase in the number of devices in the future wireless networks, these periodic updates may become difficult and may lead to loss in performance for the users. Also, the performance of the system depends on the accuracy of the position updates as well. To accommodate the increase in number of devices as well as the accuracy of the position updates, the research has focussed on predicting the future position of the users, and on developing mobility models so as to predict the movement of the users in the network.

To predict positions, multiple different methods have been proposed, from traditional algorithms such as Kalman filter based prediction schemes, to novel data driven models such as machine learning, deep learning and reinforcement learning based prediction schemes. In [9], a Deep Reinforcement Learning (DRL) algorithm for unsupervised location prediction is presented. The algorithm models the position of the Internet of Things (IoT) devices as a Markov Decision Process (MDP) and trains a DRL agent using RSSI gathered from the IoT devices to predict unknown position of the devices. In [10], the position of users is determined based on the phase, time of arrival, angle of arrival and signal strength of the



mmWave signals. It also presents an idea of using a machine learning algorithm to learn the signal strength, time of arrival and angle of arrival values and predict the location of the users. Following that, in [11], a map based position estimation mechanism is presented based on the attributes similar to those used in the previous work, i.e., angle of arrival and time of arrival, but also adds a multipath component which exploits the multipath nature of mmWave signals in the environment. Based on these information, a map of the environment is created by the system and the location of the user is determined based on the values obtained for the considered attributes. In [12], the authors use a temporal convolutional networks to determine the position estimates based on mmWave beam tracking. The temporal convolutional networks takes into account the mmWave beam signals from the user over a period of time and provides position estimates based on the previous position and trajectory. In [13], the position estimates are provided using Kalman filters to determine the position and trajectory of the vehicles in the network and find the optimal next hop for routing. In [14], each node calculates an area of coverage based on the probability of existence for the node to be in its coverage range. In [15], the authors created a dataset based on the RSSI and a random mobility model and trained a deep neural network to predict the future position of the users based on the RSSI. In [16], a dual Hidden Markov Model (HMM) is proposed. The first HMM predicts the position based on the WiFi RSSI and provides that as an input to the second HMM to determine the optimal Access Point (AP) based on the position estimate. Other approaches for the position and location prediction in different types of networks can be found in the surveys [17, 18].

### 3.1.2. Trajectory Prediction

Trajectory prediction is important as it enables the system to determine future positions of the user and optimize the system based on this information. To determine trajectory information about a user, the system has to take into account past positions over a period of time and predict the trajectory based on those positions. In [19], a Gaussian Mixed Model (GMM) is proposed which takes into account the position information over time from Global Positioning System (GPS) coordinates and determines the trajectory. It also incorporates the noise in the position information by modeling the position with a Gaussian Distribution. A maximum likelihood algorithm is used estimate the motion based on the position information and a Gaussian Process Regression (GPR) is used to predict the trajectory of the vehicle based on the trajectory data. In [20], two Recurrent Neural Networks (RNNs), Long Short-Term Memory (LSTM) and Echo State Network (ESN), are used to determine the trajectory for predicting resource allocation. In [21], user movement is predicted based on the gosition to determine the trajectory of the users. In [22], a LSTM is proposed to predict future position and using dead reckoning method to predict the trajectory based on the predicted position.

### 3.1.3. Uncertainties in Predictions

The prediction algorithms discussed previously are not always accurate, i.e., uncertainties exist in the prediction algorithms. So it is important to take that into account, especially when accurate predictions can be very crucial for optimizing the network. In [13], the location prediction is determined by Kalman filter. But Kalman filter predictions have uncertainty. To tackle that, a mean squared error is calculated to determine the quality of the prediction.



In [23], the uncertainty is predicted using deep neural networks in wireless fingerprinting based on dead reckoning. In [24], the uncertainty in three dimensional localization, based on RSSI, Time Of Arrival (TOA), Time Difference of Arrival (TDOA), Angle of Arrival (AOA) and Symmetric Double Sided Two Way Ranging (SDS-TWR), is calculated based on sensitivity factor determined by partial differential equations. Based on the uncertainty calculated, the propagation of uncertainty in the position from each of the factors is determined.

### **3.2. Communication Link Context**

Communication Link Context refers to the contextual information regarding the physical and abstract attributes of the wireless medium. Primarily, communication link refers to the channel over which the communication takes place, which is a physical attribute, as well as the quality of the channel or link, which is an abstract attribute based on the physical attribute. The link context information is crucial in wireless communication due to highly dynamic nature of wireless channels which leads to performance variations that are critical in a network containing large number of devices.

### 3.2.1. Channel State Prediction

To determine the channel state, the system needs to either gather continuous CSI updates from the user. But continuous updates increase the overhead especially for a network containing a large number of devices. In [25], the CSI is obtained for the uplink channel by the system and then used for uplink to downlink mapping to predict the channel state in the downlink channel. To predict the channel state, a deep neural network called SCNet is proposed which takes into account the uplink channel state for an Frequency Division Duplexing (FDD) Multiple Input Multiple Output (MIMO) system and predicts the downlink channel state. In [26], an LSTM is proposed to predict the next CSI based on the previous CSI. The LSTM structure is trained on the dataset created by gathering CSI obtained from a Rayleigh fading channel and then used to predict the next channel states based on previous channel states in a rolling prediction fashion (previous predictions are used to determine the next predictions). In [27], a combination of Convolutional Neural Network (CNN) and LSTM called OCEAN is used to predict CSI. A combination of frequency band, location, time, temperature, humidity, and weather conditions is used to train the network offline (historical data) and online (current data) and provides an online prediction for the channel state. In [28], a combination of Feed Forward Neural Network (FNN) and Radial Basis Function Neural Network (RBF-NN) is used to predict the channel parameters. The network is trained using an input of Transmitter (Tx) and Receiver (Rx) coordinates, Tx-Rx distance and carrier frequency, which are determined using real time measurements and a Geometry based Stochastic Model (GBSM). In [29], an LSTM is used to predict the channel state. The training dataset for the LSTM model is gathered by using the IEEE802.11p in-phase and quadrature-phase signals to gather the channel information using algorithms such as down sampling, frame detection, symbol alignment, frequency offset correction and training sequence extraction. The LSTM model is trained using the channel information obtained and then used to predict channel quality in vehicular networks.



### 3.2.2. Link State Prediction

Link State Prediction is crucial especially when determining the link guality to calculate metrics for data transmission so as to achieve an optimal network performance. In [30], a routing metric is proposed based on link quality estimation based on RSSI and Average Link Quality (ALQ), which is determined based on past measurements. Based on the link estimation, the next hop is chosen for routing in vehicular ad-hoc networks. In [31], the physical link is modelled based on the transmission energy consumption and a Mixed Integer Programming (MIP) optimization problem is presented which is solved by a deep neural network. In [32], a prediction algorithm is presented that anticipates link quality metric such as Expected Transmission Count (ETX) and Expected Transmission Time (ETT) to optimise the Optimized Link State Routing (OLSR) protocol. The link quality metric is predicted based on signal strength, which in turn is predicted based on linear regression. In [33], the Crosslayer Interference and Delay Aware Metric (CL-IDA) is estimated using multiple different machine learning techniques, such as multiple linear regression, support vector regression and Gaussian regression, and used as a metric for OLSR. In [34], a link quality state model is presented and based on the link quality model, a Kalman filter based link reliability prediction algorithm called Radio Link Reliability Prediction (RLRP) is presented which predicts the bounds of the reliability of existence of the link for the network.

### 3.3. Traffic Context

Traffic Context refers to the data traffic that is encountered in the wireless networks. Accurately predicting traffic demands in the network can be really crucial to satisfy the QoS and QoE requirements of the users, especially in a network containing a large number of devices.

### 3.3.1. Traffic Prediction

In [35], a traffic prediction algorithm called Coca-Predict based on traffic correlation and causality is presented. The algorithm uses correlation between traffic flows over time and over different cells in the area as well as traffic flows due to mobility and social events to predict the traffic at a particular place and time. The algorithm uses Auto Regression Integrated Moving Average (ARIMA), Deep Neural Network (DNN), Gradient Boosting Decision Tree (GBDT) and LightGBM [36]. In [37], an exponential smoothing is used to predict the traffic generated by the nodes in each duty cycle in Wireless Sensor Network (WSN). In [38], a Deep Belief Network and Gaussian Model (DBNG) is used to predict the traffic. The DBNG is used to predict the network traffic, which is modelled based on the previous traffic information and fluctuations in a Wireless Mesh Network (WMN). In [39], a Spatial-Temporal Cross-domain Neural Network (STC-Net) is presented which is trained via transfer learning, by using traffic information from clusters in the cellular network. In [40], a combination of a CNN and a parameter estimator is used to predict city-wide traffic. The parameter estimator is based on the distance between two cells in the cellular network of the city. In [41], a Gaussian Process based predictor is presented, which takes live data from the radio access networks and predicts traffic for a load-aware management tasks.



### 3.3.2. Performance Analysis based on Traffic Analysis

Traffic prediction is not the only thing that can be considered to improve the system. Prediction of the performance for a certain type of traffic is also necessary. Performance prediction provides a good indication of the network configurations for a certain type of traffic. This helps in determining the network configurations that can be used in different scenarios. In [42], throughput is predicted based on the maximisation of the two local and global optimization problems. The local and global optimization problems address the traffic local to the user and as a whole in the network respectively. The throughput prediction is used to devise a scheduling based on Carrier Sense Multiple Access (CSMA) threshold. In [43], a channel throughput predictor is devised which takes into account the video streaming demand of the users in the cellular network and caches information at the nearest edge to obtain the required QoE. The predictor is devised as a optimization problem that takes into account channel utilization and buffer occupancy of the user to calculate the maximum throughput. In [44], a throughput predictor is designed based on device level data such as application and Operating System (OS) information and network level data such as channel quality and cell load in cellular network. The throughput predictor is designed using three different machine and deep learning techniques namely random forest, SVM and LSTM. In [45], a machine learning based prediction algorithm is presented to predict average throughput for video streaming applications. To predict the average throughput, the RSRP, RSRQ values for primary and neighbouring cell and SNR and CQI. In [46], the throughput prediction is designed for IEEE 802.11 networks. The predictor takes into account the relationship between two metrics which is devised from interference offered by multiple access points using directional antennas and omni-directional antennas.

### 3.4. Social Context

Social Context refers to the social information about the user in the wireless network. Social information contains the information regarding user interests and content preferences, mobility trends for the users and concentration of user-defined content in the network.

### 3.4.1. Social Context based caching

Caching refers to the temporary storage of content at the edge of the network so as to reduce load on the core networks for multiple redundant requests of the same content. Social context-based caching is necessary as it stores information locally, based on the user interests and other geographical and network specific aspects such as location and mobility of the users in the network. In [47], Proactive Caching-based Mobility Prediction (PCMP) is proposed, which predicts the nearest Road Side Unit (RSU) for caching information based on user content requests and mobility in Vehicular Ad Hoc Networks (VANETs). The predictor is designed using an LSTM that takes into account the mobility information and predicts the next RSU and proactively caches information requested by the user. In [48], a logical topology is defined by combining mobile users based on their behaviours so as to create a mobile ad-hoc cloud network and optimize the network delay for computing operations. In [49], a RL scheme is proposed to take into account the mobility in VANET. A deep Q algorithm is used to devise the optimal caching and computing allocation scheme based



on the mobility of the users in the network. In [50], a social prediction algorithm using HMM based on user behaviour in social network is presented. Based on the user behaviour predictions, local cache is defined which stores the content based on the popularity predicted on the user behaviours.

### 3.4.2. Social Context based Mobility Prediction

Sometimes the system is not able to obtain mobility information from the users. To address such cases, a social context, such as previous interactions of the users with the system can be used to devise the mobility of the users. In [51], a Short Message Service (SMS) based mobility prediction scheme is defined by using SMS data to determine the location of the user and predict the mobility using two different algorithms, a MC model and Naive Bayes (NB) model. In [47], as discussed earlier, an LSTM based mobility prediction scheme is proposed. The proposed Proactive Caching-based Mobility Prediction (PCMP) predicts the nearest RSU for caching information based on user content requests and mobility in VANETs. Additionally, in [52], a survey of different mobility prediction models based on social interaction in urban areas is presented.

### 3.4.3. Social Context based Access Control

Social context can be considered to devise efficient access control and resource allocation schemes. Information such as social mobility in the network can be used to determine demand at different times of the day and year, so as to efficiently optimize the system. In [49], as discussed previously, an RL scheme is proposed to take into account the mobility in VANET. A deep Q algorithm is used to devise the optimal caching and computing allocation schemes based on the mobility of the users in the network. In [53], an ARIMA model is used to take into account user behaviour in Device to Device (D2D) networks and predict an optimal resource allocation strategy. It also takes into account the social correlations between users so as to optimize the overall network utility in a social community, i.e., a pool of users who have similar interests and behaviours. Additionally, in [54], an overall survey of socially aware resource allocation scheme in D2D networks is presented.



# 4. Anticipatory Techniques

This chapter will provide information and discuss the recent works on anticipatory networking, which apply different techniques to forecast further networking information based on contextual information obtained from network.

We have divided these techniques into three main categories and discussed them in details. These categories include: 1) *time series prediction*, that predict future values based on previously observed values; 2) *Probabilistic forecasting* methods, that make statements about the likelihood of the future events, based on available information; 3) *Data-driven* approaches, such as clustering, classification and regression that learn to make predictions by only relaying on data. In the following sections, we discussed these methods and their applications in networking and communications.

### 4.1. Time Series Prediction

The complex varying nature of wireless channel has been a subject of study of researchers for many years. Knowing how the channel behave can provide plentiful competitive advantages in terms of radio resource management, allowing for instance network operators to expand services for more users using the same amount of network resources while also maintaining strict QoS. Power control, resource blocks allocation, coding rate, transmit antennas, precoding codeword and constellation size are examples of transmission parameters that can be adjusted according to the prior knowledge of the channel condition, especially for applications that demand critical network requirements, such as Industry 4.0, tactile communication and Internet of Things.

To achieve this potential, we need to transmit signals in a closed-loop manner, receiving the CSI as a feedback from the receiver. This information can be understood as a time series that represents an estimation of the channel condition, describing the combined effect of, for instance, scattering, fading, and power decay. Nevertheless, this estimation might not be always precise. Either in frequency-division duplex (FDD) or time-division duplex (TDD) systems, problems of estimation accuracy are still raised, especially in high mobility or high frequency scenarios. This can be a serious shortcoming, notably when it comes 5G and beyond 5G network requirements, which are expected to increasingly explore the use of millimetre waves under high speed scenarios, such as autonomous vehicles and aerial navigation systems.

As precisely summarized in [55] and [56], many authors have been extensively demonstrating the impact of CSI inaccuracy over a wide variety of adaptive transmissions techniques, such as MIMO systems [57–59], beam-forming [60, 61], antenna selection [62], mobility management [63, 64], precoding [65], interference coordination [66, 67] and relaying [68, 69]. In this regard, many authors have been proposing solutions to cope outdated CSI issues [56], which can be mostly categorized into three different classes:

- 1. sub-optimal methods [70], where imperfect CSI is assumed and only part of the full performance potential can be achieved;
- 2. passive methods [71], in which wireless resources (frequency, power, time, etc) are used to passively compensate the performance loss;



3. channel prediction methods [72], in which the main idea is trying to forecast the channel behavior in advance (future CSI values), without using any additional wireless resource.

In the past few years, prediction methods have been gaining increasingly more attention from researchers due to their impact on performance and efficient use of radio resources [55]. More in details we can distinguish the prediction methods according to two different models: autoregressive (AR) [73] and parametric (PR) [74]. Both represent classical approaches to establish a statistical modeling of the wireless channel. The PR models rely on estimating fading channel parameters (e.g., Doppler shift, delay spread, angles of arrival, etc), which is a process that can demand high computational complexity and be vulnerable to changes at the scenario. The AR models, in turn, are based on an autoregressive process to extrapolate future CSI values by linearly combining past and current CSI measures. Besides its simplicity compared to the PR approach, the AR models are very susceptible to additive noise [75] and not really deployed in practice.

More recently, an alternative methodology using Machine Learning (ML) has been raising as a promising candidate to enhance prediction methods [55]. Its biggest advantage is that the data-driven nature of ML can replace statistical modeling of the channel, thus avoiding all the drawbacks involving parameters estimation and additive noise handling. In this regard, many authors have been exploring ML techniques in different prediction tasks. In [76] and [77], the authors applied LSTM networks to tackle classical channel estimation problems, including high speed scenarios. In [78], the authors propose a CNN, a technique typically used in computer vision tasks, to extract complex CSI features from the channel and predicting CSI aging. A solution for MIMO channel prediction problem was proposed by the authors in [79] and [80] using RNNs. The authors in [81] propose a novel CSI prediction scheme for improving the performance of massive MIMO, Non-Orthogonal Multiple Access (NOMA), Coordinated Multipoint (CoMP), and physical layer security by applying multi-hidden layer neural network. The work proposed by the authors in [82] provides a very insightful application of LSTM and CNN models for predicting the CSI values in downlink channel by using information from uplink channel. Likewise, the authors in [83] extrapolates the downlink CSI values from observed uplink CSI information, but also considering both Single-Input Single-Output (SISO) and MIMO scenarios. This new proposed scheme actually outperforms the classic Wiener filter-based approach.

### 4.2. Probabilistic Forecasting

*Probabilistic forecasting* methods employ available information of system to make statements about the likelihood of the future events. In the following subsections we introduce two well-known statistical methods for probabilistic forecasting: *Markovian Models* and *Bayesian inference*.

### 4.2.1. Markovian Models

A Markov model is a stochastic model used to represent random changing systems in which future states only depend on the current state. This section will provide information about Markov Chain (MC), and its applications in anticipatory networking.



A (Discrete-time) MC is a sequence of random variables  $X_1$ ,  $X_2$ ,  $X_3$ , ..., where, probability of moving to the next state, is independent from the previous states and only depends on the current state (Markov property):

$$P(X_{n+1} = x | X_1 = x_1, X_2 = x_2, ..., X_n = x_n) = P(X_{n+1} = x | X_n = x_n).$$
(4.1)

MCs can be represented by a state-flow diagram, i.e., a directed graph in which the nodes indicate the states and the edges specify the probability of moving from one state to another. Fig. 4.1 shows a classical example of MC that models the "weather" process with three possible states: {*Sunny*, *Rainy*, *Snowy*}. As an example consider the observed state in the present time step is *Sunny*, then the probability that MC takes value *Sunny*, *Rainy*, *Snowy*, in the next time step, is 0.8, 0.19 and 0.01, respectively. The sum of all probability values of the outward edges from any state is equal to 1.



Figure 4.1: Example of state flow diagram of a MC.

Having the Markov transition probabilities, and the stationary distribution of the MC states could help to predict the behavior and the state of the system in future time steps. Markov modes have already been applied in wireless communication networks, mostly for user and vehicle positioning and mobility forecasting.

Several MC-based approaches have been proposed to predict the mobility and trajectory of locations of vehicles [84] and users [85,86] in VANETs [84], Heterogeneous Cloud Radio Access Networks (HCRANs) [87], and WSNs [88], employing different models such as Hidden Markov Model (HMM) [89], second-order Markov [90], and temporal Markov models [91]. The predicted mobility information can be used for different purposes like access point or service node selection [92,93], routing in VANETs [94] or WSNs [88,95] and traffic management [96]. To achieve more accurate results, some works have jointly applied Markov models with other techniques such as classification [97] and LSTM [98]. The Spatio-temporal



correlation that existed in users' mobility patterns has also been investigated in some literature [99–101]. The method in [99] takes into account the non-Gaussian characteristics and Spatio-temporal correlation of real human mobility data to present a Markov model for the human mobility prediction.

The authors of [100] proposed a hybrid MC model that can adaptively apply first or secondorder MC to predict the future location of users, based on the quality of the mobility traces. They further presented the Zone of Interest discovery scheme in urban areas, utilizing the Spatio-temporal analysis. A HMM-based mechanism, namely CityTracker, is presented in [101]. It predicts the individual's trajectory and analyzes the representative citywide crowd mobility. The proposed algorithm can achieve a representative crowd mobility visualization in the target area by integrating individual trajectories. Markov models have also been applied in prediction of other networking parameters like as signal fading [102], energy [103], node status in WSNs [104] and indoor positioning [53]. A distributed probabilistic approach based on Hidden Gaussian Markov Model (HGMM) presented in [105], to predict sensor failure in WSNs.

#### 4.2.2. Bayesian Inference

Bayesian inference is a statistical method that applies the Bayes theorem to update the probability of a hypothesis given new information. In fact, the Bayesian inference derives the *posterior probability*, given the *prior probability* of hypothesis and a *likelihood function* applying the Bayes' theorem:

$$p(\theta \mid y) = \frac{p(y \mid \theta) \cdot p(\theta)}{p(y)}, \tag{4.2}$$

where *y* and  $\theta$  represent the data and the *hypothesis* whose probability may be affected by data.  $p(\theta)$  indicates the prior probability (probability of  $\theta$  before *y* data is observed).  $p(\theta|y)$  is the posterior probability (probability of  $\theta$ , given *y*).  $p(y|\theta)$ , called *likelihood*, represents the probability of *y* given  $\theta$ .

Recently, Bayesian inferences have been applied to predict different networking parameters from mobility [106] to channel gain [107] and reliability [108]. The work in [106] employed Bayesian inference to devise a mobility prediction model for WSN. The authors of [109] presented a feature-based Bayesian method for content requests and popularity prediction in edge-caching networks. The authors of [108] proposed a service prediction model and applied the Bayesian network method to learn and predict the reliability of the mobile wireless network. A channel gain prediction method for mobile users that exploits the Spatio-temporal correlation in a Bayesian framework is presented in [107]. Traffic prediction is another application that has been considered in some recent works. A prediction algorithm is proposed in [110] based on the Bayesian Spatio-temporal model to predict the spatial distribution of traffic in the cellular network at different moments. In [111], a wireless traffic prediction is presented based on Bayesian seasonal adjustment.

### 4.3. Clustering, classification, and Regression for Context Acquisition

Clustering, classification, and regression are three of the most frequently used techniques for extracting useful information and predicting incidents in different applications. Context ac-



quisition for wireless networks optimization is a field in which we can use the strategies mentioned above to extract information and optimize the resources based on that knowledge. In this section, we discuss a few selected methods in two main categories: similarity-based approaches and regression analysis.

### 4.3.1. Similarity-Based Approaches

The main goal in similarity-based approaches is to learn a similarity function between objects to reveal the similar latent structures in a dataset. They have a great variety of use cases in different applications, including ranking, recommendation systems, face verification, and speaker verification. These techniques have also been applied to anticipatory networking literature, which will be briefly discussed in the following, with reference to three main categories.

### 4.3.1.1. Collaborative Filtering (CF)

In collaborative filtering, which has been widely applied to recommendation systems, the underlying assumption is that if two people have the same opinion on an issue, they are more likely to have similar opinions on other issues. Based on this assumption, the CF techniques try to predict a user's opinion on an issue that is not rated by the user, given the preferences of other users.

Since there are a few comprehensive surveys on CF [112–114], we briefly introduce the concepts of CF, related to anticipatory networking in this section. In CF techniques, we usually have a matrix of size  $n_u \times n_c$  in which  $n_u$  is the number of users, and  $n_c$  is the number of contents. The goal is to predict the missing values in this matrix.

There are two major categories of CF techniques: memory-based and model-based. In the memory-based methods, similar users are identified using similarity metrics like cosine similarity or Pearson correlation, and then the missing values are predicted using a weighted average over the ratings of the similar users on the contents.

Model-based approaches are another broad group of methods for CF in which machine learning algorithms including K-Nearest Neighbors (KNN), Singular Value Decomposition (SVD), Matrix Factorization (MF) and nArtificial Neural Network (ANN) are used to estimate the ratings of the unrated contents in the matrix.

Network caching is one of the critical concepts in ensuring QoE to the users. J. Yao et al. have recently published a thorough survey on caching strategies and techniques on mobile edge [115], in which several techniques have used different types of collaborative filtering. Huda S. Goian *et al.* have also published another comprehensive survey on caching-enabled networks from a popularity-based video caching perspective [116]. In this survey, there are a few works on neural-network-based CF. Another work proposes a collaborative multicast beamforming approach for content delivery in cache-enabled networks [117] in which the popularity of contents are estimated and, based on that, users are served using the cached content in Small Base Stations (SBSs).

Similar to popularity-based content caching, several works have proposed approaches to use users' locations to provide them with the proper content, including [118].



### 4.3.1.2. Clustering

Clustering is a broad group of unsupervised machine learning techniques that aims to identify different subsets of objects in a dataset in the way that the objects inside a group are similar to each other.



Figure 4.2: Example of k-means algorithm on a synthetic dataset after six steps with k equals to 4.

In this section, we briefly introduce the k-means clustering algorithm as one of the most frequently used clustering algorithms in anticipatory networking. We are given a dataset  $x^{(1)}, ..., x^{(m)}$  with *m* points. Each of these points has a feature vector  $x^{(i)} \in \mathbb{R}^n$  of dimension *n*. The goal is to assign group labels  $c^i = 1, 2, ..., K$ , to each point in the dataset in such a way that the points in the same group are close with respect to a distance metric defined over the feature space.

The k-means algorithm is described in Algo. 1, and a step by step visualization of the algorithm on a synthetic dataset with k = 4 is shown in Fig. 4.2.

Algorithm 1: K-means Clustering Algorithm

1 initialize **cluster centroids**  $\mu_1, \mu_2, ..., \mu_k \in \mathbb{R}^n$  randomly ; 2 **while** *the* **centroids** are not stabilized **do** 3 for every *i*, set  $c^i := \arg \max_j ||x^{(i)} - \mu_j||^2$  ; 4 for each *j*, set  $\mu_j := (1/||c^j||) \sum_{i=1}^{||c^j||} x^i$ , where  $||c^j||$  represents the number of data points in the *j*<sup>th</sup> cluster 5 **end** 



Different clustering algorithms, including k-means, have been applied to a broad range of cases in anticipatory networking. Optimizing the energy consumption in WSNs is one of the challenging problems for which a few clustering-based approaches have been proposed [119, 120].

Vehicular scenarios is another field in which clustering techniques play a significant role. To guarantee the stability of the self-organized communication structure in multimedia communication on the Internet of Vehicles (IoV), Kai Lin *et al.* propose a content-aware model [121]. In another work in the vehicular domain, [122], Ashit K. Dutta *et al.* introduce a hierarchical clustering protocol to optimize the resource utilization in the network and increase the overall lifetime. Aligned with the previous work, MCA-V2I [123] is a multi-hub clustering mechanism aiming to decrease the number of control messages and increase the stability by using a master/slave paradigm in IoV.

### 4.3.1.3. Decision Tree-Based Classification

Decision tree learning approaches are simple yet powerful models in ML either for classification or for regression tasks. They have a tree-like structure consisting of decision nodes and leaves. Decision nodes direct the data toward a proper leave, and the leaves are the labels of the classification task.

This simple ML method has been applied to a vast number of problems, and anticipatory networking is not an exception. Decision tree-based methods play an essential role in localization and positioning systems using Radio Frequency(RF) signals. Chanama *et al.* have studied different decision tree-based methods in indoor positioning systems [124]. Another group of researchers has proposed a None Line of Sight (NLoS) location tracking system based on decision trees that use Ultra Wide Band (UWB) technology [125].

Decision trees have also been considered for device type classification in WSNs [126], multiobject detection and classification in outdoor scenarios [127], and relay selection for dualhop wireless communications [128].

### 4.3.2. Regression Analysis

Regression analysis is a powerful statistical tool for examining the relationship between a few independent variables with a dependent variable of interest. In anticipatory networking, a great variety of papers have used regression analysis for different tasks. In this section, we introduce three of the most useful regression techniques and their use cases in anticipatory networking.

### 4.3.2.1. Support Vector Machine

SVM [129] is a discriminative classifier defined by a decision hyperplane separating instances of different classes with the maximum margin. There is a regression version of SVM named Support Vector Regression (SVR) sharing the same principles with SVM but for regression tasks. In this section, we briefly introduce SVMs and overview recently published works in anticipatory networking using SVM and SVR.

We explain SVM in a binary classification scenario which can be extended to multiple classes classification tasks too. Imagine we have a training dataset  $\{(x_i, y_i) | x_i \in \mathbb{R}^n, y_i \in \{-1, 1\}, i =$ 



1,2,..., M}, where  $x_i$  is the  $i^{th}$  training sample in an n dimensional space,  $y_i$  is the corresponding label for that sample, and M is the total number of training samples.

We assume that the samples are linearly separable, thus there is a hyperplane  $w \cdot x - b = 0$  (where  $W \cdot x$  is the inner product of W and x), which separates those two classes of samples. The optimization problem that yields decision hyperplane with the largest distance from the support vectors (the closest vectors to the decision hyperplane) is:

$$\begin{array}{ll} \min_{w,b} & \frac{1}{2} ||w||^2 \\ \text{s.t.} & y_i(w^T x_i + b) \ge 1, i = 1, 2, ..., M. \end{array}$$
(4.3)

This optimization problem is not able to find an optimal hyperplane for non-linear decision boundaries. To solve this problem, kernel methods have been proposed. In kernel-based SVM instead of using the original input attributes x, we can transform the original input using a feature mapping function  $\Phi$  and then apply the SVM. Since the equation (4.3) can be entirely written in terms of the inner product  $x_i \cdot x_j$ , we can simply replace  $x_i \cdot x_j$  with  $\Phi(x_i) \cdot \Phi(x_j)$ . The linear model in the new space corresponds to a non-linear model in the original space.



(a) Linear kernel on a linearly non-separable synthetic dataset. (b) The data after applying a Gaussian kernel to add a new dimension *z*.

(c) Classification with SVM with Gaussian kernel on a linearly non-separable synthetic dataset.

Figure 4.3: Examples of applying the SVM classifier with a linear and Gaussian kernels on a linearly non-separable synthetic dataset.

One of the most frequently used kernels in the literature is the Gaussian kernel:  $\exp(-\frac{||x_i-x_j||^2}{2\sigma^2})$ . An example of applying a linear and Gaussian kernel-based SVM classifier on a non-linear synthetic data is shown in Fig. 4.3. When we apply the Gaussian kernel, the data become linearly separable in a space with more dimensions which is equal to a non-linear decision boundary in the original space. A recent and comprehensive survey on SVMs and their applications is [130]. Additionally, SVM attempts to find a decision boundary for the given supervised set while SVR attempts to find a curve that fits the supervised set.

B. Silva *et al.* proposed a machine learning-based context prediction system to boost vehicle-to-cloud communication [131]. In this work they have used SVR to predict the data rate by feeding three sets of features: mobility context, channel context, and application



context to the model. Another group of researchers has applied a non-linear SVR to cancel the self-interfering signal in full-duplex communication systems [132]. They have used two separate SVR models for predicting the real and the imaginary parts of the interfering signal. Coverage area detection [133] and network quality prediction [134–136] have also benefited from SVRs for building a coverage map and predicting the data rate in vehicular scenarios to improve the resource efficiency and the network reliability, respectively.

### 4.3.2.2. Artificial Neural Networks

Artificial Neural Networks (ANNs) are universal function approximators consisting of neurons grouped in multiple layers (input, hidden, and output layers). ANNs can be used for different tasks including clustering, classification and regression. Each neuron receives the input vector X from the previous layer, multiplies them by a vector of weights W, applies a summation over them, adds a bias term b, and finally uses an activation function (Sigmoid, Tanh, ReLU, ELU, etc.) g to introduce non-linearity:

$$\hat{y} = g(W^T \cdot X + b). \tag{4.4}$$

To train an ANN, the forward and a backward pass is used. In the forward pass, a chain rule is applied to determine the gradient and in the backward pass, a back propagation mechanism is used to adjust the weights and biases of the layers by using an optimizer such as Adam, RMSProb, Gradient Descent, Momentum, etc. Using this technique, we try to minimize the loss function to update learning parameters of the network. A loss function is dependent on the optimization problem to be solved and the dataset used for training the ANN. With the advance of computational resources like Graphics Processing Units (GPUs) and Tensor Processing Units (TPUs) and abundance of data, especially in networking, ANNs have increasingly gained attention in the literature.

CSI estimation is one of the key challenges in wireless communications. OCEAN [27] is an online neural network model that uses CNN and LSTM to predict CSI in 5G wireless communication systems. SCNet [25] is another ANN model that aims to predict the downlink CSI in Base Station (BS) of a FDD MIMO communication system using an autoencoder-like design which not only reduces the redunduncy of data but also makes it more robust to noise.

Additionally, Millimeter Wave (mmWave) communication is a relevant component of 5G and beyond. Beam selection and blockage prediction are two of the most important challenges in these systems. M. Alrabeiah et al. [137] have proposed a deep learning-based approach to tackle these challenges using ResNet18 [138] architecture.

Context-aware wireless communication optimization is another direction to which a few ANNbased approaches have been applied. J. Ren et al. have proposed a deep learning model using CNN to extract fine-grained features and the applied SVM classifier to classify the applications in a wireless network to improve QoE [139]. Another group of researchers have used ANNs as well as other classifiers like SVMs for classifying the Wireless Local Area Network (WLAN) traffic to reduce the power consumption [140].



## 5. Wireless Network Optimization Techniques using Antic-

### ipatory and Context Aware Techniques

This chapter will focus on optimization techniques based on anticipatory networking solutions. From a high-level prospective, we have divided these techniques in two main categories. First, we discussed traditional approaches to solve convex optimization problems, and latter, we introduced different methods for solving non-convex optimization problems. These methods include: Model Predictive Control (MPC), Reinforcement Learning (RL), Deep Neural Network (DNN) and Game theory, that is introduced in the rest of this chapter.

### 5.1. Optimization Approaches for Convex Optimization Problem

In general, a mathematical optimization problem has the form of

minimize 
$$f_0(x)$$
  
s.t.  $f_i(x) \le b_i, i = 1, ..., m$  (5.1)

where  $x = (x_1, ..., x_n)$  is a vector called *optimization variable* of the problem,  $f_0 : \mathbb{R}^n \to \mathbb{R}$  is the *objective function*,  $f_i : \mathbb{R}^n \to \mathbb{R}$ , i = 1, ..., m are the *constraint functions*, and  $b_1, ..., b_m$  are the *bounds* for the constraints. A vector  $x^*$  is called optimal, if for any z that satisfies all the constraints, we have  $f_0(z) \ge f_0(x^*)$ .

We generally define different *classes* or *families* of optimization problems based on the form of the objective and constraint functions. For example, an optimization problem is called *Linear Program (LP)* if the objective and the constraint functions are linear, i.e.,

$$f_i(\alpha x + \beta y) = \alpha f_i(x) + \beta f_i(y)$$
(5.2)

for all  $x, y \in \mathbb{R}^n$  and for all  $\alpha, \beta \in \mathbb{R}$ . LP has more specific types: Integer Linear Program (ILP) in which the optimization variable vector is integer and Mixed-Integers Linear Program (MILP) in which the optimization variable vector is a mixture of integer and real values.

A specific and very important type of optimization classes is called *convex optimization*, which is a generalization of the LP. A optimization problem, is said to be convex when the objective and constraint functions are all convex, i.e.,

$$f_i(\alpha x + \beta y) \le \alpha f_i(x) + \beta f_i(y)$$
(5.3)

for all  $x, y \in \mathbb{R}^n$ , and for all  $\alpha, \beta \in \mathbb{R}$ , with  $\alpha + \beta = 1, \alpha \ge 0, \beta \ge 0$ .

Resource allocation in anticipatory networking is usually modeled as a LP, ILP, MILP or a convex optimization problem, in which the resources are considered as the optimization variables and the prediction is modeled through the constraints of the problem [1]. N. H. Bidoki *et al.* have proposed a two-objective LP approach to optimize the Value of Information (VoI) of the data being sent by sensors over a wireless communication network and energy consumption [141]. In the first objective function they try determine the paths and sleep schedule of each node while maximizing the VoI of the sensed data. In the second problem, they focus on minimizing the overall energy consumption of the network.



Another power allocation scheme is investigated in an integrated radar-communication system [142]. The authors have formulated the target detection problem as an LP optimization. They have shown that the low probability of intercept-based optimal power allocation performance can be improved by using their approach.

Another group of researchers have proposed a secrecy rate optimization scheme to increase the physical layer's security and increase the spectrum efficiency through full-duplex wire-less communication [143]. First, they design a mixed integer and non-linear programming paradigm, they then relax this problem into a LP through reformulation-linearization technique.

A. Alsharoa *et al.* proposed a MILP optimization-based approach in an Unmanned Aerial Vehicle (UAV) scenario to optimize the transmit power levels and trajectories of the relaying UAVs with the aim of maximizing the data rate transmission of the ground users with an undirect link [144]. In particular, they have proposed a two-step approach. In the first step, they have utilized MILP to optimally determine the users-UAVs associations and UAVs' transmit power levels. In the the second step, a recursive method based on shrink-and-realign process is presented to optimize the UAVs trajectories.

### 5.2. Optimization Approaches for Non-Convex Optimization Problem

### 5.2.1. Model Predictive Control

Model Predictive Control (MPC) [145, 146] is an advanced process control method that optimizes a sequence of actions in dynamic systems by solving a finite horizon open-loop optimal control problem. The MPC technique has been widely applied in multi-variable constrained control problems. However, because of its mathematical complexity and large computational cost, it is only practical for processes with slow dynamics.

MPC has been recently applied to solve some problems in wireless communications. The work in [147] takes advantage of MPC to present a holistic approach for Virtual Machine (VM) placement in cloud data centers by considering conflicting performance metrics, such as counteracting hardware outages or software aging issues, security policies, perceived service level, and power requirements. An explicit MPC method is presented in [148] to reduce the energy consumption of a Fully Electric Vehicle (FEV) by predicting the preceding vehicle movements. A congestion control framework for infrastructure-based Cognitive Radio Networks (CRNs) is presented in [149] where an active queue management algorithm based on multiple MPC is proposed. In [150], the authors define a decentralized MPC mechanism for traffic control in freeways by considering a lossy communication channels. An autonomous IoT model predictive controller for linear multi-agent systems in the presence of imperfect network is presented in [151] to solve the data losses in communication channels. The authors of [152] applied MPC to co-design control and routing of Wireless Sensor and Actuator Networks (WSANs).

The authors of [153] used MPC to present an energy-efficient communication model for relay-assisted UAV networks. Some works applied the MPC to provide collision avoidance in UAV systems. A collision avoidance system for UAVs in civil arispace is presented in [154] that applies a distributed reactive MPC for trajectory tracking. A hierarchical collision avoidance system based on online path planning, distributed MPC, and geometric reactive



control approaches for multiple fixed-wind UAVs is presented in [155]. The effects of the collision avoidance mechanism on the mobility of UAV swarms is investigated in [156].

### 5.2.2. Reinforcement Learning

Reinforcement Learning (RL) is a machine learning approach for sequential decision making problems in which an agent learns to interact with the environment in an action-reward loop. The environment is modeled as a Markov Decision Process (MDP), which is represented by  $(S, A, P, r, \gamma)$  where S defines different states of the environment, A is a finite set of possible actions (action space) and P is transition probability matrix where the element  $p_{ss'}$  determines the probability of transition from state s to s' in a step. The term r indicates the reward function  $r : S \times A \times S \rightarrow \mathbb{R}$  which represents the reward that the agent obtains by moving from state s to s' by taking action a. The term  $\gamma$  is a discount factor used to the trade-off between immediate and long-term rewards and for mathematical tractability of non-episodic environments. As shown in Fig. 5.1, in each time step t the agent observes a state  $s_t$  and takes the action  $a_t$ . The environment reacts to the action, moves to the new state  $s_{t+1}$ , and returns the reward  $r_{t+1}$  to the agent. The ultimate goal for a RL algorithm is to maximize the cumulative future reward.



Figure 5.1: Action-reward loop in RL

The behavior of an RL agent in state *s* is described by a policy  $\pi(s)$ , which is defined as a probability distribution over the action space. The agent learns to optimize the policy to increase the reward obtained over time.

The objective of training an agent is to find the optimal policy  $\pi^*$ , which tells the agent which action to take in a certain state. So, by starting from  $s_t$  the goal of the agent is to maximize the expected reward (return) until the end of episode. This goal is summarised as follows:

$$G_t = \mathbb{E}\left[\sum_{k=0}^{\infty} \gamma^k r(s_{t+k}, a_{t+k}) | s_0 = s_t\right]$$
(5.4)

### 5.2.2.1. Q-Learning

For most real-world applications, the transition probability  $\mathcal{P}$  is unknown, and thus, it is difficult to estimate the policy  $\pi^*$ . Hence, to provide reliable policy estimations, Q-learning can be used. Q-learning is a famous off-policy (against the on-policy and actor-critic) algorithm that utilizes a lookup table, known as Q-table, to store Q-values for all action-state pairs. A



Q-value for (*s*, *a*) pair under a policy  $\pi$  tells us how much is good being in state *s* and taking action *a* and thereafter following policy  $\pi$ . The optimal Q-table can be obtained by randomly initializing and iteratively updating it by using the Bellman Equation:

$$Q^{*}(s, a) = r(s, a) + \gamma \max_{a_{t+1}} Q(s_{t+1}, a_{t+1}).$$
(5.5)

To avoid local optima and create a balance between exploration and exploitation, Q-learning adopts the  $\epsilon$ -greedy algorithm, where the agent can exploit its previous knowledge by selecting action  $a = \arg \max_{a' \in A} Q(s, a')$  with probability  $1 - \epsilon$  or explore the environment by selecting a random action with probability  $\epsilon$ .

RL has recently attracted a lot of attention in the communication community, and many researchers have modeled wireless systems as a MDP and applied RL techniques to tackle them. Q-learning has recently been considered in several papers, but here we will discuss only on a few of the most recent and significant ones. In [157], the authors presented a Q-learning based method to maximize the spectral efficiency in multi-service wireless systems. Channel selection problem for cognitive radio networks is considered in [158]. A Multi-objective optimization routing protocol for flying Ad Hoc networks, exploiting Q-learning is presented in [159]. In [160], a Q-learning based algorithm proposed to control the sleep mode in 5G networks adaptively.

The main issue with the tabular setting of the Q-learning algorithm is that it assumes the states are discrete, which is very uncommon in real-world applications. Most real-world systems have a huge and continuous state space that is impossible to represent in a tabular manner. One solution, known as Deep Q-Network (DQN) or more generally deep RL consists in approximating Q using DNNs as discussed in the next section.

### 5.2.2.2. Deep Reinforcement Learning

In DQN, the *Q* function is approximated by minimizing the squared error between Bellman equation and the DNN estimation:  $(Q(s_t, a_t; \theta) - Q^{\text{target}})^2$ , where  $\theta$  is the DNN parameters, and  $Q^{\text{target}}$  is the target *Q* function given by:

$$Q^{\text{target}} = r(s_t, a_t) + \gamma \max_{a_t + a} Q(s_{t+1}, a_{t+1}, \theta)$$
(5.6)

The use of experience replay memory and target-network makes the learning process more stable [161].

Introducing the DQNs, made it possible to apply RL in many real world problems, including in wireless communication. A comprehensive survey on applications of DRL in communication networks is provided by [162–164]. Some other surveys exist in the literature that investigate the DRL papers in a certain topic. The authors in [165] discussed the application of DRL for resource management in 5G Heterogeneous Networkss (HetNets). The DRL protocols for VANETs, is considered in [166], while dynamic spectrum allocation based on RL in cognitive radio networks is presented in [167].

Some optimization frameworks based on RL are considered in recent works. In [168] the authors introduced unified model-based and model-free reinforced-unsupervised frameworks, to solve the variable and functional optimization problems. They employed unsupervised learning for the known model of the environment while incomplete, they used reinforcement learning algorithms to learn the model.



DRL-based resource management methods for wireless networks has been widely considered in recent researches [169], for soliving different problems like traffic scheduling in roadside communications [170], cellular network traffic [171], and 5G Medium Access Control (MAC) layer [172], or power control for spectrum sharing in Cognitive Radios (CRs) [173]. DRL. The Age of Information (AoI)-aware Radio Resource Management (RRM) in Vehicle to Vehicle (V2V) communication network is investigated in [174], by modeling the stochastic decision-making procedure as a discrete-time single-agent MDP. They proposed a proactive algorithm based on LSTM and DRL techniques to address the partial observability and overcome the curse of dimensionality. A DRL-based framework is presented in [175, 176] to provide model-free Ultra Reliable Low Latency Communication (URLLC) in downlink of Orthogonal Frequency Division Multiple Access (OFDMA) systems. In particular [175] proposed a framework to allocate power and Resource Blocks (RBs) to users, considering their end-to-end reliability and latency without any modeling assumption of traffic or packet sizes. In [176], the authors propose a new Generative Adversarial Networks (GANs)-based approach to pre-train the DRL agent on mixed real and synthetic dataset to expose the agent to a broad range of network conditions.

Resource allocation is a classical control problem in wireless communication that modeled as MDP and solve by RL in some works. Some papers presented model-driven DRL assisted resource allocation for ultra-dense cellular networks [177, 178]. The authors of [179] presented a traffic and Channel-aware OFDMA resource allocation, while incremental RL for dynamic resource allocation is investigated in [180]. A circumstance-independent resource allocation method is presented in [181] to address the different network circumstances with a single policy. An actor-critic-based RL framework for dynamic channel access presented in [182]. Some other recent works considered resource allocation in vehicular networks [183–186]. As an example, a resource allocation method is presented in [186], where the base station applies graph-based techniques to allocate channels in a centralized manner, and the vehicles employ DRL to realize distributed power control. One can find a survey Multi-agent RL methods for vehicular networks in [187].

The QoS, as an important networking parameters is considered in some RL frameworks. A QoS-constrained resource allocation for multiservice networks is presented in [188]. DRL is employed in [189] to present a power control scheme for quality-driven video transmission. A power-efficient resource allocation framework for cloud Radio Access Networks (RANs) is presented in [190]. The authors of [191] modeled the predictive power allocation for mobile video streaming as a DRL problem and resorted to Deep Deterministic Policy Gradient (DDPG) to learn the proper policy. A predictive power allocation mechanism for video streaming over mobile networks with deep RL is presented in [191, 192].

DRL is also a hot topic in HetNets and most of the papers applied DRL techniques to jointly control and optimize different networking parameters. As a matter of fact, a joint dynamic channel access and power control using DRL is considered in [193] while [194] considered the joint power control and user association in mmWave HetNets. A joint user association and resource allocation optimization is considered by the authors in [195, 196], while [197] is investigated the user association optimization jointly with power control. An self-optimization framework based on contextual bandit is presented in [198] that jointly applies energy control and interference coordination.

The RL is also considered in adaptive Modulation and Coding Scheme (MCS) selection in cognitive HetNets [199, 200] and high throughput wireless networks [201]. Some newer pa-



pers investigated the possibility of employing hierarchical and multi-agent RL framework in wireless systems. The paper in [202] proposed a multi-agent RL for dynamic power allocation. A hierarchical distributed optimization algorithm for multi-agent reinforcement learning is presented in [203]. RL has recently been applied in some modern communication problems like resource management in network slicing [204–206] or Intelligent Reflecting Surface (IRS) optimization in Multiple-input Single-output (MISO) communication systems [207, 208].

### 5.2.3. Optimization Approaches based on Data Driven Model Prediction

Data driven approaches have shown a significant success in a wide range of applications and wireless communication optimization is not an exception. The abundance of data in wireless networking make it a suitable field to apply data driven models to optimize different characteristics of the network from energy efficiency to spectrum sharing.

Haibo He *et al.* propose a deep reinforcement learning based framework to optimize the energy efficiency for distributed cooperative spectrum sharing [209]. They formulate the energy efficiency problem as a combinatorial optimization problem and integrate the graph neural networks and deep reinforcement learning techniques to optimize it. The role of the graph neural network in their architecture is to embed the sensor network's graph-structured data into a feature vector of a fixed length. Due to the unavailability of the optimal solution, they have used a deep reinforcement learning approach to optimize the problem.

A. Zappone *et al.* proposed an interesting approach for wireless optimizations [210]. By embedding the prior expert knowledge into a neural network architecture, they decreased the amount of data needed for the training of the model. They applied the proposed approach to the energy-efficiency optimization problem in wireless communication. In the power allocation optimization domain, another deep-learning enabled online energy-efficient mechanism has been proposed [211]. They have designed an ANN architecture to receives the network communication channels and estimate the power allocation to each user.

Another group of researchers have utilized a data driven optimization method to optimize detection in terms of Bit Error Rate (BER), which in turn, enables reliable communication over Additive White Gaussian Noise (AWGN) and multi-path Rayleigh fading channels with bandwidth constraints [212]. They have used auto-encoders to estimate the received signal and a fully-connected layer to detect the symbol.

System capacity and service coverage trade-off in cellular wireless communication networks is another optimization problem for which a deep-learning enabled approach called DECCO is proposed [213]. In this work, a data driven optimization mechanism is used to derive group alignment of user signal strength and minimum Signal-To-Interference-plus-Noise (SINR) during coverage and capacity optimization.

### 5.2.4. Optimization Approaches based on Game theory

Game theory was originally developed for analyzing the interactive decision processes, predicting the outcomes of interactions, and identifying optimal strategies in economics. The selfish nature of wireless communication systems in which each User Equipments (UE) competes with other UEs to obtain resources to maximize its spectral efficiency and QoE makes it a suitable domain for applying game theory techniques [214].



Game theory based wireless communication optimization techniques fall into three main categories: one-to-one (each UE is assigned to a single resource and vice versa), one-to-many (each UE is assigned to a single or multiple resources and vice versa), many-to-many (multiple UEs are assigned to multiple resources and vice versa).

T. Liu *et al.* [215] propose a game theory based approach in caching-enabled heterogeneous cellular networks to minimize UEs' transmission latency and maximize QoE. They decouple the problem into a two-stage matching problem. In the first stage they define a many-to-many matching problem of matching Small Base Stations (SBSs) who have cached certain files to UEs (SBSs-UEs), and in the second stage they formulate the problem as a many-to-one matching game of associating each Service Provider (SP) with SBSs (SP-SBSs). Another group of researchers have studied the problem of user clustering and power assignment in a D2D communication with NOMA scheme in wireless cellular networks using a two-stage approach [216]. In the first stage, they define two one-to-many matching games to associate each cellular and D2D user with a cluster (sub-channel) and in the second stage they propose a Successive Convex Approximation (SCA)-based with Arithmetic-Geometric Mean (AGM) approximation approach for power allocation.


# 6. Applications of Optimization Techniques in Wireless Networks

Optimization techniques in wireless networks have been used to address a variety of applications across the stack. In this chapter, we discuss the applications of such optimization techniques based on anticipatory information from the wireless network.

## 6.1. Physical Layer

In the current literature, we can find numerous optimization techniques applied to the network PHY layer. In [217], the authors investigate the problem of how to maximize the sum rate of a 2-user uplink mm-wave-NOMA system, proposing a sub-optimal solution where the original problem is divided into two sub-problems: one is a power control and beam gain allocation problem, and the other is a beamforming problem under the constant-modulus (CM) constraint. The authors in [218] propose a centralized as well as a distributed power control algorithm, aiming to maximize the capacity of a D2D network. They consider a scenario of licensed and unlicensed spectrum, showing by simulations that the proposed approach could increase the throughput of the D2D networks compared with current state-of-the-art methodologies. In [219], the authors address the energy efficiency optimisation problem of a NOMA 5G wireless network. The proposed idea is based on improving energy efficiency of 5G terminals while satisfying the constraints on maximum transmit power budget, minimum data rate, and minimum harvested energy per terminal. The proposed scheme is compared with an exhaustive search method, showing convergence to a stable optimal value.

The authors in [220] propose a new approach for multiple access in the fifth generation (5G) of cellular networks called power domain sparse code multiple access (PSMA). They compare the PSMA with other proposed NOMA strategies from the perspective of receiver complexity and system performance, showing that PSMA significantly outperforms other NOMA techniques while imposing a reasonable increase in complexity to the system, considering both aspects of transmitter and receiver.

We can check in [221] a flexible mechanism for Discontinuous Reception (DRX) proposed for 5G networks, where the goal is to minimize the energy usage of user devices for applications of video streaming while preventing buffer underflows. The authors show that the proposed approach can provide an energy reduction usage by up to 60% compared to static DRX setups.

The work in [222] proposes a novel channel state feedback scheme, where the minimum number of feedback bits is calculated with respect to the channel reconstruction error. The authors compare the proposed approach with other traditional feedback schemes, showing analytically and numerically the advantages of the proposed solution, especially in conditions of low SNR.

A pilot placement optimization problem is proposed by the authors in [223] for the radio access in 5G vehicle-to-everything communications to support Internet of Vehicles applications. The authors formulate the problem as a MDP, aiming to find enhanced pilot patterns for assisting OFDM technology. Simulation results show that the proposed scheme was able to effectively track fast time-varying vehicular channels.



In [224], the authors provide an optimization algorithm for calculating the joint maximumlikelihood estimation of channel and clipping level at the receiver side of IoT-based OFDM networks. These network scenarios are typically characterized for having lots of low-cost low-power transmitters and more complex receiver nodes, such as a base station. Numerical evaluations show that the proposed estimator was capable to achieve almost the same performance of a perfect estimator, where both channel and clipping level are perfectly known.

#### 6.2. MAC Layer

At MAC layer, a main focus has been on applying optimization techniques for radio resource allocation and error correction. In [225], the authors propose a novel scheduling framework that is able to select different scheduling policies according to the states of the scheduler by using RL principles. The main goal is minimizing packet delay and packet loss rates for attending the strict 5G network requirements. Simulation results show that the proposed scheme outperforms traditional packet schedulers.

Likewise, the authors in [226] also propose a scheduling framework based on a RL strategy. Their idea is to optimize QoS provision management for heterogeneous traffic. The proposed scheme is capable of maximizing the average scheduling time when heterogeneous QoS requirements are met for diverse traffic classes, achieving a performance up to 50% higher than state-of-the-art solutions.

The work proposed in [227] provides a solution for the ultra-reliable low-latency communications (URLLC) and enhanced mobile broadband (eMBB) coexistence on the same radio spectrum. The authors introduce a novel scheduler framework that aims at optimizing a cross-objective function, where the critical URLLC QoS is guaranteed while extracting the maximum possible eMBB ergodic capacity. Simulation results show that the proposed scheme guarantees instant scheduling for sporadic URLLC traffic, and with minimal impact on the overall ergodic capacity, overcoming state-of-the-art scheduling proposals from both industry and academia.

In [228], the authors optimize the hybrid automatic repeat request (HARQ) operation of 5G URLLC networks. The problem is formulated according to a non-convex and mixed integer programming, which is analytically intractable. The authors came up with a solution based on optimizing a repetition coding scheme, and showing, by simulation results, minimization o the required bandwidth to achieve URLLC traffic requirements.

Likewise, the authors in [229] also propose a solution for optimizing HARQ operation in order to achieve URLLC requirements. The proposed scheme applies the queuing delay model based on the Pollaczek-Khinchine (P-K) formula, aiming to optimize bandwidth considering HARQ of 5G URLLC communication. Analytical and simulation results show that the proposed mechanism was able to reduce bandwidth requirements by up to 64.8% compared to traditional approaches.

The work established in [230] utilizes Tennessee Eastman (TE) process model to create a novel concept for communication-edge-computing (CEC) loop, introducing an optimization problem for achieving the defined CEC efficiency with focus on industrial IoT applications. The authors derive a new uplink (UL)-based communication protocol, that was able to outperform typical HARQ performance in terms of latency, reliability, and bandwidth efficiency.



## 6.3. Network Layer

In Network Layer, the optimization techniques are mainly applied to routing problems. In [13], a location error resilient routing scheme is proposed for VANETs. It predicts the location of the vehicles in the network using Kalman filter and creates a routing directory to decide the next forwarding node in the network. The protocol is designed to optimize the selection of the next forwarding node to maximise the throughput and minimise the routing load in the network.

In [231], a Genetic algorithm-based energy-efficient clustering and routing algorithm (GECR) is presented which takes into account the location of the nodes and devise routing and clustering strategies in WSN. The cluster head selection and next hop forwarding node is devised by modelling the optimization problem and solving it using a genetic algorithm. The algorithm aims to maximise the network time and minimise the energy consumption in the network. Similarly in [232], a multi objective pareto optimization approach is proposed to solve the problem of clustering and routing while taking into account energy efficiency, reliability and scalability.

In [233], an optimization problem called minimum broadcast power is discussed and is solved using a hybrid particle swarm optimization algorithm (H-PSO). The H-PSO aims at minimising the transmission power so as to conserve energy in broadcast mode of transmission in wireless ad-hoc network.

In [234], a non supervised deep learning based routing algorithm is proposed to optimize traffic load in wireless Software Defined Networks (SDNs). The SDN controller trains CNN to take into account traffic information and network performance such as path delay and make routing decisions based on these experiences.

In [235], a multihop routing scheme based on classification using machine learning techniques for Mobile Ad-hoc Networks (MANETs) is presented. In this scheme, the algorithm takes into account the battery power utilization and internal storage of a node to determine the next hop and eventually the whole route. To do so, each neighbouring node is classified using three classification techniques namely Multinomial Logistic Regression (MLR), SVM and KNN are used.

#### 6.4. Application Layer

In Application Layer, the optimization techniques are generally applied to performance predictions over wireless network for applications. In [43], an adaptive HTTP video streaming approach is discussed which takes into account the performance variations due to the dynamic nature of wireless networks. The system predicts the channel throughput and maximises the average download segment level or video quality while taking into account the channel utilization and end user buffer. Similarly, in [45], a throughput prediction scheme is proposed which also takes into account channel information to optimize and improve the video streaming quality.

In [236], a multi-stage ML is presented by combining an unsupervised feature extraction with a supervised classifier to extract the *quality-rate* characteristic of unknown videos that can be used in QoE-aware video admission controls and resource management. In [237],



In [238], a flow rate allocation algorithm over Multipath Transmission Control Protocol (MP-TCP) is defined called Energy Distortion Aware MP-TCP. The protocol aims to minimising the energy consumption while maximising video quality. The optimization problem takes into account the channel quality, delay, and video streaming requirement to provide an optimized flow allocation and energy consumption.

In [239], a cross layer UDP based protocol which takes into account the wireless links to provide a guaranteed performance. It basically optimizes functions such as in order delivery, forward error correction and flow control so as to minimise delay and overhead in the wireless network.



## 7. References

- [1] N. Bui, M. Cesana, S. A. Hosseini, Q. Liao, I. Malanchini, and J. Widmer, "A survey of anticipatory mobile networking: Context-based classification, prediction methodologies, and optimization techniques," *IEEE Communications Surveys & Tutorials*, vol. 19, no. 3, pp. 1790–1821, 2017.
- [2] W. Wanalertlak, B. Lee, C. Yu, M. Kim, S.-M. Park, and W.-T. Kim, "Behavior-based mobility prediction for seamless handoffs in mobile wireless networks," *Wireless Networks*, vol. 17, no. 3, pp. 645–658, 2011.
- [3] F. Beister and H. Karl, "Predicting mobile video inter-download times with Hidden Markov Models," IEEE 10th International Conference on Wireless and Mobile Computing, Networking and Communications (WiMob), pp. 359–364, 2014.
- [4] A. Noulas, S. Scellato, N. Lathia, and C. Mascolo, "Mining user mobility features for next place prediction in location-based services," *IEEE 12th international conference on data mining*, pp. 1038–1043, 2012.
- [5] S. J. Nawaz, S. K. Sharma, S. Wyne, M. N. Patwary, and M. Asaduzzaman, "Quantum Machine Learning for 6G Communication Networks: State-of-the-Art and Vision for the Future," *IEEE Access*, vol. 7, pp. 46317–46350, 2019.
- [6] B. Gyires-Tóth, P. Varga, and T. Tóthfalusi, "Utilizing Deep Learning for Mobile Telecommunications Network Management," *IFIP/IEEE Symposium on Integrated Network and Service Management (IM)*, pp. 575–580, 2019.
- [7] R. Falkenberg, B. Sliwa, N. Piatkowski, and C. Wietfeld, "Machine learning based uplink transmission power prediction for LTE and upcoming 5G networks using passive downlink indicators," *IEEE 88th Vehicular Technology Conference (VTC-Fall)*, pp. 1–7, 2018.
- [8] N. Bui and J. Widmer, "Data-driven evaluation of anticipatory networking in LTE networks," *IEEE Transactions on Mobile Computing*, vol. 17, no. 10, pp. 2252–2265, 2018.
- [9] Y. Li, X. Hu, Y. Zhuang, Z. Gao, P. Zhang, and N. El-Sheimy, "Deep reinforcement learning (DRL): Another perspective for unsupervised wireless localization," *IEEE Internet of Things Journal*, 2019.
- [10] O. Kanhere and T. S. Rappaport, "Position locationing for millimeter wave systems," in 2018 IEEE Global Communications Conference (GLOBECOM), pp. 206–212, 2018.
- [11] O. Kanhere, S. Ju, Y. Xing, and T. S. Rappaport, "Map-assisted millimeter wave localization for accurate position location," in 2019 IEEE Global Communications Conference (GLOBECOM), pp. 1–6, IEEE, 2019.
- [12] J. Gante, G. Falcão, and L. Sousa, "Deep Learning Architectures for Accurate Millimeter Wave Positioning in 5G," *Neural Processing Letters*, vol. 51, no. 1, pp. 487–514, 2020.



- [13] R. Kasana, S. Kumar, O. Kaiwartya, W. Yan, Y. Cao, and A. H. Abdullah, "Location error resilient geographical routing for vehicular ad-hoc networks," *IET Intelligent Transport Systems*, vol. 11, no. 8, pp. 450–458, 2017.
- [14] U. Palani, K. Suresh, and A. Nachiappan, "Mobility prediction in mobile ad hoc networks using eye of coverage approach," *Cluster Computing*, vol. 22, no. 6, pp. 14991– 14998, 2019.
- [15] Y. Yayeh, H.-p. Lin, G. Berie, A. B. Adege, L. Yen, and S.-S. Jeng, "Mobility prediction in mobile ad-hoc network using deep learning," in 2018 IEEE International Conference on Applied System Invention (ICASI), pp. 1203–1206, IEEE, 2018.
- [16] K.-L. Yap, Y.-W. Chong, and W. Liu, "Enhanced handover mechanism using mobility prediction in wireless networks," *PloS one*, vol. 15, no. 1, p. e0227982, 2020.
- [17] L. N. Balico, A. A. Loureiro, E. F. Nakamura, R. S. Barreto, R. W. Pazzi, and H. A. Oliveira, "Localization prediction in vehicular ad hoc networks," *IEEE Communications Surveys & Tutorials*, vol. 20, no. 4, pp. 2784–2803, 2018.
- [18] H. Zhang and L. Dai, "Mobility prediction: A survey on state-of-the-art schemes and future applications," *IEEE Access*, vol. 7, pp. 802–822, 2018.
- [19] L.-I. Wang, Z.-g. Chen, and J. Wu, "Vehicle trajectory prediction algorithm in vehicular network," *Wireless Networks*, vol. 25, no. 4, pp. 2143–2156, 2019.
- [20] W. Zhang, Y. Liu, T. Liu, and C. Yang, "Trajectory prediction with recurrent neural networks for predictive resource allocation," in 2018 14th IEEE International Conference on Signal Processing (ICSP), pp. 634–639, IEEE, 2018.
- [21] X. Liu, Y. Liu, Y. Chen, and L. Hanzo, "Trajectory design and power control for multi-UAV assisted wireless networks: A machine learning approach," *IEEE Transactions* on Vehicular Technology, vol. 68, no. 8, pp. 7957–7969, 2019.
- [22] Q. Liu, G. Chuai, J. Wang, and J. Pan, "Proactive Mobility Management With Trajectory Prediction Based on Virtual Cells in Ultra-Dense Networks," *IEEE Transactions on Vehicular Technology*, vol. 69, no. 8, pp. 8832–8842, 2020.
- [23] Y. Li, Z. Gao, Z. He, Y. Zhuang, A. Radi, R. Chen, and N. El-Sheimy, "Wireless fingerprinting uncertainty prediction based on machine learning," *Sensors*, vol. 19, no. 2, p. 324, 2019.
- [24] X. Yan, Q. Luo, P. Zhou, and J. Liu, "An Uncertainty Propagation Mechanism Analysis Method for Three-dimensional Quadrilateral Localization," in 2019 Prognostics and System Health Management Conference (PHM-Paris), pp. 39–44, IEEE, 2019.
- [25] Y. Yang, F. Gao, G. Y. Li, and M. Jian, "Deep learning-based downlink channel prediction for FDD massive MIMO system," *IEEE Communications Letters*, vol. 23, no. 11, pp. 1994–1998, 2019.



- [26] G. Liu, Y. Xu, Z. He, Y. Rao, J. Xia, and L. Fan, "Deep learning-based channel prediction for edge computing networks toward intelligent connected vehicles," *IEEE Access*, vol. 7, pp. 114487–114495, 2019.
- [27] C. Luo, J. Ji, Q. Wang, X. Chen, and P. Li, "Channel state information prediction for 5G wireless communications: A deep learning approach," *IEEE Transactions on Network Science and Engineering*, 2018.
- [28] J. Huang, C.-X. Wang, L. Bai, J. Sun, Y. Yang, J. Li, O. Tirkkonen, and M. Zhou, "A big data enabled channel model for 5G wireless communication systems," *IEEE Transactions on Big Data*, 2018.
- [29] J. Joo, M. C. Park, D. S. Han, and V. Pejovic, "Deep learning-based channel prediction in realistic vehicular communications," *IEEE Access*, vol. 7, pp. 27846–27858, 2019.
- [30] K. N. Qureshi, F. Bashir, and A. H. Abdullah, "Distance and signal quality aware next hop selection routing protocol for vehicular ad hoc networks," *Neural Computing and Applications*, vol. 32, no. 7, pp. 2351–2364, 2020.
- [31] A. Akbas, H. U. Yildiz, A. M. Ozbayoglu, and B. Tavli, "Neural network based instant parameter prediction for wireless sensor network optimization models," *Wireless Networks*, vol. 25, no. 6, pp. 3405–3418, 2019.
- [32] S. Naimi, A. Busson, V. Vèque, and R. Bouallegue, "Metric anticipation to manage mobility in mobile mesh and ad hoc wireless networks," *Annals of Telecommunications*, vol. 73, no. 11-12, pp. 787–802, 2018.
- [33] M. Naravani, D. Narayan, S. Shinde, and M. M. Mulla, "A Cross-Layer Routing Metric with Link Prediction in Wireless Mesh Networks," *Procedia Computer Science*, vol. 171, pp. 2215–2224, 2020.
- [34] W. Sun, Q. Li, J. Wang, L. Chen, D. Mu, and X. Yuan, "A radio link reliability prediction model for wireless sensor networks," *International Journal of Sensor Networks*, vol. 27, no. 4, pp. 215–226, 2018.
- [35] J. Wen, M. Sheng, J. Li, and K. Huang, "Assisting Intelligent Wireless Networks with Traffic Prediction: Exploring and Exploiting Predictive Causality in wireless traffic," *IEEE Communications Magazine*, vol. 58, no. 6, pp. 26–31, 2020.
- [36] G. Ke, Q. Meng, T. Finley, T. Wang, W. Chen, W. Ma, Q. Ye, and T.-Y. Liu, "Lightgbm: A highly efficient gradient boosting decision tree," in *Advances in neural information* processing systems, pp. 3146–3154, 2017.
- [37] H. Luo, M. He, Z. Ruan, and F. Chen, "A duty-cycle mac algorithm with traffic prediction for wireless sensor networks," in 2017 4th International Conference on Information Science and Control Engineering (ICISCE), pp. 16–19, IEEE, 2017.
- [38] L. Nie, D. Jiang, S. Yu, and H. Song, "Network traffic prediction based on deep belief network in wireless mesh backbone networks," in 2017 IEEE Wireless Communications and Networking Conference (WCNC), pp. 1–5, IEEE, 2017.



- [39] C. Zhang, H. Zhang, J. Qiao, D. Yuan, and M. Zhang, "Deep transfer learning for intelligent cellular traffic prediction based on cross-domain big data," *IEEE Journal on Selected Areas in Communications*, vol. 37, no. 6, pp. 1389–1401, 2019.
- [40] C. Zhang, H. Zhang, D. Yuan, and M. Zhang, "Citywide cellular traffic prediction based on densely connected convolutional neural networks," *IEEE Communications Letters*, vol. 22, no. 8, pp. 1656–1659, 2018.
- [41] Y. Xu, F. Yin, W. Xu, J. Lin, and S. Cui, "Wireless traffic prediction with scalable Gaussian process: Framework, algorithms, and verification," *IEEE Journal on Selected Areas in Communications*, vol. 37, no. 6, pp. 1291–1306, 2019.
- [42] A. Adel Aly, H. M ELAttar, H. ElBadawy, and W. Abbas, "Aggregated Throughput Prediction for Collated Massive Machine-Type Communications in 5G Wireless Networks," *Sensors*, vol. 19, no. 17, p. 3651, 2019.
- [43] L. Tao, Y. Gong, S. Jin, and J. Zhao, "Energy-efficient HTTP Adaptive Streaming with Anticipated Channel Throughput Prediction in Wireless Networks," in *Proceedings of* the 20th ACM International Conference on Modelling, Analysis and Simulation of Wireless and Mobile Systems, pp. 253–258, 2017.
- [44] D. Raca, A. H. Zahran, C. J. Sreenan, R. K. Sinha, E. Halepovic, R. Jana, and V. Gopalakrishnan, "On Leveraging Machine and Deep Learning for Throughput Prediction in Cellular Networks: Design, Performance, and challenges," *IEEE Communications Magazine*, vol. 58, no. 3, pp. 11–17, 2020.
- [45] D. Raca, A. H. Zahran, C. J. Sreenan, R. K. Sinha, E. Halepovic, R. Jana, V. Gopalakrishnan, B. Bathula, and M. Varvello, "Empowering video players in cellular: throughput prediction from radio network measurements," in *Proceedings of the 10th ACM Multimedia Systems Conference*, pp. 201–212, 2019.
- [46] S. Kandasamy, R. Morla, P. Ramos, and M. Ricardo, "Predicting throughput in IEEE 802.11 based wireless networks using directional antenna," *Wireless Networks*, vol. 25, no. 4, pp. 1567–1584, 2019.
- [47] H. Khelifi, S. Luo, B. Nour, A. Sellami, H. Moungla, and F. Naït-Abdesselam, "An optimized proactive caching scheme based on mobility prediction for vehicular networks," in 2018 IEEE Global Communications Conference (GLOBECOM), pp. 1–6, IEEE, 2018.
- [48] C. Li, Z. Liye, T. Hengliang, and L. Youlong, "Mobile user behavior based topology formation and optimization in ad hoc mobile cloud," *Journal of Systems and Software*, vol. 148, pp. 132–147, 2019.
- [49] R. Q. Hu *et al.*, "Mobility-aware edge caching and computing in vehicle networks: A deep reinforcement learning," *IEEE Transactions on Vehicular Technology*, vol. 67, no. 11, pp. 10190–10203, 2018.
- [50] E. J. Kitindi and G. Rehman, "User Aware Edge Caching in 5G Wireless Networks," *IJCSNS*, vol. 18, no. 1, p. 25, 2018.



- [51] R. Fernandes and R. D. GL, "A new approach to predict user mobility using semantic analysis and machine learning," *Journal of medical systems*, vol. 41, no. 12, p. 188, 2017.
- [52] J. Wang, X. Kong, F. Xia, and L. Sun, "Urban human mobility: Data-driven modeling and prediction," *ACM SIGKDD Explorations Newsletter*, vol. 21, no. 1, pp. 1–19, 2019.
- [53] Z. Deng, W. Liu, X. Yan, W. Wang, *et al.*, "Multisimi-Markov: An Improved Markov Position Prediction Method," in *2018 Ubiquitous Positioning, Indoor Navigation and Location-Based Services (UPINLBS)*, pp. 1–7, IEEE, 2018.
- [54] E. Ahmed, I. Yaqoob, A. Gani, M. Imran, and M. Guizani, "Social-aware resource allocation and optimization for D2D communication," *IEEE wireless communications*, vol. 24, no. 3, pp. 122–129, 2017.
- [55] W. Jiang and H. D. Schotten, "Neural network-based fading channel prediction: A comprehensive overview," *IEEE Access*, vol. 7, pp. 118112–118124, 2019.
- [56] W. Jiang and H. D. Schotten, "Deep Learning for Fading Channel Prediction," *IEEE Open Journal of the Communications Society*, vol. 1, pp. 320–332, 2020.
- [57] M. Li, M. Lin, W.-P. Zhu, Y. Huang, A. Nallanathan, and Q. Yu, "Performance analysis of MIMO MRC systems with feedback delay and channel estimation error," *IEEE Transactions on Vehicular Technology*, vol. 65, no. 2, pp. 707–717, 2015.
- [58] Q. Wang, L. J. Greenstein, L. J. Cimini, D. S. Chan, and A. Hedayat, "Multi-user and single-user throughputs for downlink MIMO channels with outdated channel state information," *IEEE Wireless Communications Letters*, vol. 3, no. 3, pp. 321–324, 2014.
- [59] K. T. Truong and R. W. Heath, "Effects of channel aging in massive MIMO systems," *Journal of Communications and Networks*, vol. 15, no. 4, pp. 338–351, 2013.
- [60] J.-B. Kim, J.-W. Choi, and J. M. Cioffi, "Cooperative distributed beamforming with outdated CSI and channel estimation errors," *IEEE Transactions on Communications*, vol. 62, no. 12, pp. 4269–4280, 2014.
- [61] Y. Isukapalli, R. Annavajjala, and B. D. Rao, "Performance analysis of transmit beamforming for MISO systems with imperfect feedback," *IEEE Transactions on Communications*, vol. 57, no. 1, pp. 222–231, 2009.
- [62] X. Yu, W. Xu, S.-H. Leung, and J. Wang, "Unified performance analysis of transmit antenna selection with OSTBC and imperfect CSI over Nakagami-m fading channels," *IEEE Transactions on Vehicular Technology*, vol. 67, no. 1, pp. 494–508, 2017.
- [63] Y. Teng, M. Liu, and M. Song, "Effect of outdated CSI on handover decisions in dense networks," *IEEE Communications Letters*, vol. 21, no. 10, pp. 2238–2241, 2017.
- [64] L. Liang, J. Kim, S. C. Jha, K. Sivanesan, and G. Y. Li, "Spectrum and power allocation for vehicular communications with delayed CSI feedback," *IEEE Wireless Communications Letters*, vol. 6, no. 4, pp. 458–461, 2017.



- [65] J. Zheng and B. D. Rao, "Capacity analysis of MIMO systems using limited feedback transmit precoding schemes," *IEEE Transactions on Signal Processing*, vol. 56, no. 7, pp. 2886–2901, 2008.
- [66] P. Aquilina and T. Ratnarajah, "Performance analysis of IA techniques in the MIMO IBC with imperfect CSI," IEEE Transactions on Communications, vol. 63, no. 4, pp. 1259– 1270, 2015.
- [67] X. Chen and C. Yuen, "On interference alignment with imperfect CSI: Characterizations of outage probability, ergodic rate and SER," *IEEE Transactions on Vehicular Technology*, vol. 65, no. 1, pp. 47–58, 2015.
- [68] J. L. Vicario, A. Bel, J. A. Lopez-Salcedo, and G. Seco, "Opportunistic relay selection with outdated CSI: outage probability and diversity analysis," *IEEE Transactions on Wireless Communications*, vol. 8, no. 6, pp. 2872–2876, 2009.
- [69] M. Chen, T. C.-K. Liu, and X. Dong, "Opportunistic multiple relay selection with outdated channel state information," *IEEE Transactions on Vehicular Technology*, vol. 61, no. 3, pp. 1333–1345, 2012.
- [70] D. J. Love, R. W. Heath, V. K. Lau, D. Gesbert, B. D. Rao, and M. Andrews, "An overview of limited feedback in wireless communication systems," *IEEE Journal on selected areas in Communications*, vol. 26, no. 8, pp. 1341–1365, 2008.
- [71] W. Jiang, T. Kaiser, and A. H. Vinck, "A robust opportunistic relaying strategy for cooperative wireless communications," *IEEE Transactions on Wireless Communications*, vol. 15, no. 4, pp. 2642–2655, 2015.
- [72] A. Duel-Hallen, "Fading channel prediction for mobile radio adaptive transmission systems," *Proceedings of the IEEE*, vol. 95, no. 12, pp. 2299–2313, 2007.
- [73] K. E. Baddour and N. C. Beaulieu, "Autoregressive modeling for fading channel simulation," *IEEE Transactions on Wireless Communications*, vol. 4, no. 4, pp. 1650–1662, 2005.
- [74] R. O. Adeogun, P. D. Teal, and P. A. Dmochowski, "Extrapolation of MIMO mobile-tomobile wireless channels using parametric-model-based prediction," *IEEE Transactions on Vehicular Technology*, vol. 64, no. 10, pp. 4487–4498, 2014.
- [75] W. Jiang and H. D. Schotten, "A comparison of wireless channel predictors: Artificial Intelligence versus Kalman filter," in *ICC 2019-2019 IEEE International Conference on Communications (ICC)*, pp. 1–6, IEEE, 2019.
- [76] H. Ye, G. Y. Li, and B.-H. Juang, "Power of deep learning for channel estimation and signal detection in OFDM systems," *IEEE Wireless Communications Letters*, vol. 7, no. 1, pp. 114–117, 2017.
- [77] Y. Liao, Y. Hua, X. Dai, H. Yao, and X. Yang, "ChanEstNet: A deep learning based channel estimation for high-speed scenarios," in *ICC 2019-2019 IEEE International Conference on Communications (ICC)*, pp. 1–6, IEEE, 2019.



- [78] J. Yuan, H. Q. Ngo, and M. Matthaiou, "Machine learning-based channel estimation in massive MIMO with channel aging," in 2019 IEEE 20th International Workshop on Signal Processing Advances in Wireless Communications (SPAWC), pp. 1–5, IEEE, 2019.
- [79] C. Potter, K. Kosbar, and A. Panagos, "MIMO channel prediction using recurrent neural networks," International Foundation for Telemetering, 2008.
- [80] G. Routray and P. Kanungo, "Rayleigh fading MIMO channel prediction using RNN with genetic algorithm," in *International Conference on Computational Intelligence and Information Technology*, pp. 21–29, Springer, 2011.
- [81] R.-F. Liao, H. Wen, J. Wu, H. Song, F. Pan, and L. Dong, "The Rayleigh fading channel prediction via deep learning," *Wireless Communications and Mobile Computing*, vol. 2018, 2018.
- [82] J. Wang, Y. Ding, S. Bian, Y. Peng, M. Liu, and G. Gui, "UL-CSI data driven deep learning for predicting DL-CSI in cellular FDD systems," *IEEE Access*, vol. 7, pp. 96105– 96112, 2019.
- [83] M. Arnold, S. Dörner, S. Cammerer, S. Yan, J. Hoydis, and S. t. Brink, "Enabling FDD massive MIMO through deep learning-based channel prediction," *arXiv preprint arXiv:1901.03664*, 2019.
- [84] D. Papakostas and D. Katsaros, "A Rich-Dictionary Markov Predictor for Vehicular Trajectory Forecasting," in 2018 IEEE 30th International Conference on Tools with Artificial Intelligence (ICTAI), pp. 556–563, IEEE, 2018.
- [85] L. Menz, R. Herberth, C. Luo, F. Gauterin, A. Gerlicher, and Q. Wang, "An improved method for mobility prediction using a Markov Model and density estimation," in 2018 IEEE Wireless Communications and Networking Conference (WCNC), pp. 1–6, IEEE, 2018.
- [86] J. Guo, L. Liu, S. Zhang, and J. Zhu, "Mobility Prediction with Missing Locations Based on Modified Markov Model for Wireless Users," in 2019 IEEE International Congress on Big Data (BigDataCongress), pp. 132–138, IEEE, 2019.
- [87] H. Park and Y. Lim, "A Markov-based prediction algorithm for user mobility at heterogeneous cloud radio access network," in 2019 IEEE International Conference on Big Data and Smart Computing (BigComp), pp. 1–5, IEEE, 2019.
- [88] G. A. Montoya and Y. Donoso, "A prediction algorithm based on Markov chains for finding the minimum cost path in a mobile wireless sensor network," in 2018 7th International Conference on Computers Communications and Control (ICCCC), pp. 169–175, IEEE, 2018.
- [89] A. T. Akabane, R. W. Pazzi, E. R. Madeira, and L. A. Villas, "Modeling and prediction of vehicle routes based on hidden Markov Model," in 2017 IEEE 86th Vehicular Technology Conference (VTC-Fall), pp. 1–5, IEEE, 2017.



- [90] D. Zhao, Y. Gao, Z. Zhang, Y. Zhang, and T. Luo, "Prediction of Vehicle Motion Based on Markov Model," in 2017 International Conference on Computer Systems, Electronics and Control (ICCSEC), pp. 205–209, IEEE, 2017.
- [91] F. R. de Araújo, D. L. Rosário, K. Machado, E. C. Cerqueira, and L. Villas, "TEMMUS: A Mobility Predictor based on Temporal Markov Model with User Similarity," in *Anais Principais do XXXVII Simpósio Brasileiro de Redes de Computadores e Sistemas Distribuídos*, pp. 594–607, SBC, 2019.
- [92] K.-L. Yap and Y.-W. Chong, "Optimized access point selection with mobility prediction using hidden Markov Model for wireless network," in *2017 Ninth International Conference on Ubiquitous and Future Networks (ICUFN)*, pp. 38–42, IEEE, 2017.
- [93] J. Yan, C. Chen, and D. Zhang, "Node selection method based on Markov location prediction in vehicle networking," in 2018 IEEE 4th Information Technology and Mechatronics Engineering Conference (ITOEC), pp. 754–758, IEEE, 2018.
- [94] L. Yao, J. Wang, X. Wang, A. Chen, and Y. Wang, "V2X routing in a VANET based on the Hidden Markov Model," *IEEE Transactions on Intelligent Transportation Systems*, vol. 19, no. 3, pp. 889–899, 2017.
- [95] S. Nagadivya and R. Manoharan, "Energy Efficient Markov Prediction Based Opportunistic Routing (Eempor) For Wireless Sensor Networks," in 2019 IEEE International Conference on System, Computation, Automation and Networking (ICSCAN), pp. 1–6, IEEE, 2019.
- [96] H. Liu, L. T. Yang, J. Chen, M. Ye, J. Ding, and L. Kuang, "Multivariate multi-order Markov multi-modal prediction with its applications in network traffic management," *IEEE Transactions on Network and Service Management*, vol. 16, no. 3, pp. 828–841, 2019.
- [97] Y. Li, L. Lei, and M. Yan, "Mobile user Location Prediction Based on user Classification and Markov Model," in *2019 International Joint Conference on Information, Media and Engineering (IJCIME)*, pp. 440–444, IEEE, 2019.
- [98] P. Wang, H. Wang, H. Zhang, F. Lu, and S. Wu, "A hybrid Markov and LSTM model for indoor location prediction," *IEEE Access*, vol. 7, pp. 185928–185940, 2019.
- [99] Y. Qiao, Z. Si, Y. Zhang, F. B. Abdesslem, X. Zhang, and J. Yang, "A hybrid Markovbased model for human mobility prediction," *Neurocomputing*, vol. 278, pp. 99–109, 2018.
- [100] M. Karimzadeh, Z. Zhao, F. Gerber, and T. Braun, "Pedestrians complex behavior understanding and prediction with hybrid Markov chain," in 2018 14th International Conference on Wireless and Mobile Computing, Networking and Communications (WiMob), pp. 200–207, IEEE, 2018.
- [101] S.-H. Fang, L. Lin, Y.-T. Yang, X. Yu, and Z. Xu, "CityTracker: Citywide Individual and Crowd Trajectory Analysis Using Hidden Markov Model," *IEEE Sensors Journal*, vol. 19, no. 17, pp. 7693–7701, 2019.



- [102] S. KORDNOORİ, H. MOSTAFAEİ, and M. BEHZADI, "An Application of a Semi Markov Model in Predicting the Fading Occurrences in Wireless Communication Channels," *Gazi University Journal of Science*, vol. 33, no. 2, pp. 566–577.
- [103] P. Theerthagiri, "CoFEE: Context-aware futuristic energy estimation model for sensor nodes using Markov Model and autoregression," *International Journal of Communication Systems*, p. e4248, 2019.
- [104] P. Chaturvedi and A. Daniel, "Hidden Markov Model based node status prediction technique for target coverage in wireless sensor networks," in 2017 International Conference on Intelligent Communication and Computational Techniques (ICCT), pp. 223– 227, IEEE, 2017.
- [105] M. Saihi, A. Zouinkhi, B. Boussaid, M. N. Abdelkarim, and G. Andrieux, "Hidden Gaussian Markov Model for distributed fault detection in wireless sensor networks," *Transactions of the Institute of Measurement and Control*, vol. 40, no. 6, pp. 1788–1798, 2018.
- [106] F. Somaa, C. Adjih, I. El Korbi, and L. A. Saidane, "A Bayesian model for mobility prediction in wireless sensor networks," in 2016 International Conference on Performance Evaluation and Modeling in Wired and Wireless Networks (PEMWN), pp. 1–7, IEEE, 2016.
- [107] Q. Liao, S. Valentin, and S. Stańczak, "Channel gain prediction in wireless networks based on spatial-temporal correlation," in 2015 IEEE 16th International Workshop on Signal Processing Advances in Wireless Communications (SPAWC), pp. 400–404, IEEE, 2015.
- [108] Y. Hu, Q. Meng, and J. Zhu, "Bayesian Network Based Reliability Prediction in Mobile Wireless Networks," in 2019 International Conference on Communications, Information System and Computer Engineering (CISCE), pp. 299–303, IEEE, 2019.
- [109] S. Mehrizi, A. Tsakmalis, S. Chatzinotas, and B. Ottersten, "A feature-based Bayesian method for content popularity prediction in edge-caching networks," in 2019 IEEE Wireless Communications and Networking Conference (WCNC), pp. 1–6, IEEE, 2019.
- [110] Z. Zhang, F. Liu, Z. Zeng, and W. Zhao, "A traffic prediction algorithm based on Bayesian spatio-temporal model in cellular network," in 2017 International Symposium on Wireless Communication Systems (ISWCS), pp. 43–48, IEEE, 2017.
- [111] X. Wang and Q. Zhou, "Wireless Traffic Prediction Based on Bayesian Seasonal Adjustment Algorithm," in 2020 IEEE 5th International Conference on Cloud Computing and Big Data Analytics (ICCCBDA), pp. 431–435, IEEE, 2020.
- [112] X. Su and T. M. Khoshgoftaar, "A survey of collaborative filtering techniques," *Advances in artificial intelligence*, vol. 2009, 2009.
- [113] R. Chen, Q. Hua, Y. Chang, B. Wang, L. Zhang, and X. Kong, "A Survey of Collaborative Filtering-Based Recommender Systems: From Traditional Methods to Hybrid Methods Based on social networks," *IEEE Access*, vol. 6, pp. 64301–64320, 2018.



- [114] H. Ambulgekar, M. K. Pathak, and M. Kokare, "A survey on collaborative filtering: tasks, approaches and applications," in *Proceedings of International Ethical Hacking Conference 2018*, pp. 289–300, Springer, 2019.
- [115] J. Yao, T. Han, and N. Ansari, "On Mobile Edge Caching," *IEEE Communications Surveys Tutorials*, vol. 21, no. 3, pp. 2525–2553, 2019.
- [116] H. S. Goian, O. Y. Al-Jarrah, S. Muhaidat, Y. Al-Hammadi, P. Yoo, and M. Dianati, "Popularity-based video caching techniques for cache-enabled networks: a survey," *IEEE Access*, vol. 7, pp. 27699–27719, 2019.
- [117] H. T. Nguyen, H. D. Tuan, T. Q. Duong, H. V. Poor, and W. Hwang, "Collaborative Multicast Beamforming for Content Delivery by Cache-Enabled Ultra Dense Networks," *IEEE Transactions on Communications*, vol. 67, no. 5, pp. 3396–3406, 2019.
- [118] Y. Liu, A. Liu, N. N. Xiong, T. Wang, and W. Gui, "Content Propagation for Content-Centric Networking Systems From Location-Based Social Networks," *IEEE Transactions on Systems, Man, and Cybernetics: Systems*, vol. 49, no. 10, pp. 1946–1960, 2019.
- [119] S. Su and S. Zhao, "An optimal clustering mechanism based on Fuzzy-C means for wireless sensor networks," *Sustainable Computing: Informatics and Systems*, vol. 18, pp. 127–134, 2018.
- [120] Y. Padmanaban and M. Muthukumarasamy, "Energy-efficient clustering algorithm for structured wireless sensor networks," *IET Networks*, vol. 7, no. 4, pp. 265–272, 2018.
- [121] K. Lin, F. Xia, and G. Fortino, "Data-driven clustering for multimedia communication in Internet of vehicles," *Future Generation Computer Systems*, vol. 94, pp. 610–619, 2019.
- [122] A. K. Dutta, M. Elhoseny, V. Dahiya, and K. Shankar, "An efficient hierarchical clustering protocol for multihop Internet of vehicles communication," *Transactions on Emerging Telecommunications Technologies*, vol. 31, no. 5, p. e3690, 2020.
- [123] O. Senouci, Z. Aliouat, and S. Harous, "MCA-V2I: A Multi-hop Clustering Approach over Vehicle-to-Internet communication for improving VANETs performances," *Future Generation Computer Systems*, vol. 96, pp. 309–323, 2019.
- [124] L. Chanama and O. Wongwirat, "A comparison of decision tree based techniques for indoor positioning system," in 2018 International Conference on Information Networking (ICOIN), pp. 732–737, 2018.
- [125] A. Musa, G. D. Nugraha, H. Han, D. Choi, S. Seo, and J. Kim, "A decision tree-based nlos detection method for the uwb indoor location tracking accuracy improvement," *International Journal of Communication Systems*, vol. 32, no. 13, p. e3997, 2019.
- [126] M. N. Yuldashev, A. I. Vlasov, and A. N. Novikov, "Energy-efficient algorithm for classification of states of wireless sensor network using machine learning methods," *Journal* of *Physics: Conference Series*, vol. 1015, p. 032153, May 2018.



- [127] A. Ahmed, A. Jalal, and K. Kim, "Region and Decision Tree-Based Segmentations for Multi-Objects Detection and Classification in Outdoor Scenes," in 2019 International Conference on Frontiers of Information Technology (FIT), pp. 209–2095, 2019.
- [128] X. Wang, "Decision-Tree-Based Relay Selection in Dualhop Wireless Communications," *IEEE Transactions on Vehicular Technology*, vol. 68, no. 6, pp. 6212–6216, 2019.
- [129] C. Cortes and V. Vapnik, "Support-vector networks," *Machine learning*, vol. 20, no. 3, pp. 273–297, 1995.
- [130] J. Cervantes, F. Garcia-Lamont, L. Rodríguez-Mazahua, and A. Lopez, "A comprehensive survey on support vector machine classification: applications, challenges and trends," *Neurocomputing*, 2020.
- [131] B. Sliwa, R. Falkenberg, T. Liebig, N. Piatkowski, and C. Wietfeld, "Boosting vehicleto-cloud communication by machine learning-enabled context prediction," *IEEE Transactions on Intelligent Transportation Systems*, 2019.
- [132] M. Emara, K. Roth, L. G. Baltar, M. Faerber, and J. Nossek, "Nonlinear digital selfinterference cancellation with reduced complexity for full duplex systems," in WSA 2017; 21th International ITG Workshop on Smart Antennas, pp. 1–6, VDE, 2017.
- [133] H. Wang, Y. Zhou, and W. Sha, "Research on wireless coverage area detection technology for 5G mobile communication networks," *International Journal of Distributed Sensor Networks*, vol. 13, no. 12, 2017.
- [134] B. Sliwa and C. Wietfeld, "Empirical analysis of client-based network quality prediction in vehicular multi-MNO networks," in 2019 IEEE 90th Vehicular Technology Conference (VTC2019-Fall), pp. 1–7, IEEE, 2019.
- [135] B. Sliwa, R. Falkenberg, T. Liebig, J. Pillmann, and C. Wietfeld, "Machine learning based context-predictive car-to-cloud communication using multi-layer connectivity maps for upcoming 5g networks," in 2018 IEEE 88th Vehicular Technology Conference (VTC-Fall), pp. 1–7, IEEE, 2018.
- [136] B. Sliwa, T. Liebig, R. Falkenberg, J. Pillmann, and C. Wietfeld, "Efficient machine-type communication using multi-metric context-awareness for cars used as mobile sensors in upcoming 5g networks," in 2018 IEEE 87th Vehicular Technology Conference (VTC Spring), pp. 1–6, IEEE, 2018.
- [137] M. Alrabeiah, A. Hredzak, and A. Alkhateeb, "Millimeter Wave Base Stations with Cameras: Vision-Aided Beam and Blockage Prediction," in 2020 IEEE 91st Vehicular Technology Conference (VTC2020-Spring), pp. 1–5, IEEE, 2020.
- [138] K. He, X. Zhang, S. Ren, and J. Sun, "Deep residual learning for image recognition," in *Proceedings of the IEEE conference on computer vision and pattern recognition*, pp. 770–778, 2016.
- [139] J. Ren and Z. Wang, "A novel deep learning method for application identification in wireless network," *China Communications*, vol. 15, no. 10, pp. 73–83, 2018.



- [140] A. Saeed and M. Kolberg, "Towards optimizing WLANs power saving: Novel contextaware network traffic classification based on a machine learning approach," *IEEE Access*, vol. 7, pp. 3122–3135, 2018.
- [141] N. H. Bidoki, M. B. Baghdadabad, G. Sukthankar, and D. Turgut, "Joint value of information and energy aware sleep scheduling in wireless sensor networks: A linear programming approach," in 2018 IEEE International Conference on Communications (ICC), pp. 1–6, IEEE, 2018.
- [142] C. Shi, F. Wang, M. Sellathurai, J. Zhou, and S. Salous, "Low probability of interceptbased optimal power allocation scheme for an integrated multistatic radar and communication system," *IEEE Systems Journal*, vol. 14, no. 1, pp. 983–994, 2019.
- [143] F. Tian, X. Chen, S. Liu, X. Yuan, D. Li, X. Zhang, and Z. Yang, "Secrecy rate optimization in wireless multi-hop full duplex networks," *IEEE Access*, vol. 6, pp. 5695–5704, 2018.
- [144] A. Alsharoa, H. Ghazzai, M. Yuksel, A. Kadri, and A. E. Kamal, "Trajectory optimization for multiple UAVs acting as wireless relays," in 2018 IEEE International Conference on Communications Workshops (ICC Workshops), pp. 1–6, IEEE, 2018.
- [145] M. Morari and J. H. Lee, "Model predictive control: past, present and future," *Computers & Chemical Engineering*, vol. 23, no. 4-5, pp. 667–682, 1999.
- [146] L. Dai, Y. Xia, M. Fu, and M. Mahmoud, "Discrete-time model predictive control," *Advances in Discrete Time Systems*, pp. 77–116, 2012.
- [147] M. Gaggero and L. Caviglione, "Model predictive control for energy-efficient, qualityaware, and secure virtual machine placement," *IEEE Transactions on Automation Science and Engineering*, vol. 16, no. 1, pp. 420–432, 2018.
- [148] S. Zhang, Y. Luo, K. Li, and V. Li, "Real-time energy-efficient control for fully electric vehicles based on an explicit model predictive control method," *IEEE Transactions on Vehicular Technology*, vol. 67, no. 6, pp. 4693–4701, 2018.
- [149] K. Xiao, S. Mao, and J. K. Tugnait, "MAQ: A multiple model predictive congestion control scheme for cognitive radio networks," *IEEE Transactions on Wireless Communications*, vol. 16, no. 4, pp. 2614–2626, 2017.
- [150] D. Bianchi, A. Borri, M. D. Di Benedetto, and A. Ferrara, "Decentralized Model Predictive Control of Freeway Traffic Systems over Lossy Communication Networks," in 2019 IEEE International Conference on Systems, Man and Cybernetics (SMC), pp. 1074– 1079, IEEE, 2019.
- [151] R. Kouki, A. Boe, T. Vantroys, and F. Bouani, "Autonomous Internet of Things predictive control application based on wireless networked multi-agent topology and embedded operating system," *Proceedings of the Institution of Mechanical Engineers, Part I: Journal of Systems and Control Engineering*, vol. 234, no. 5, pp. 577–595, 2020.



- [152] D. Satoh, K. Kobayashi, and Y. Yamashita, "MPC-based co-design of control and routing for wireless sensor and actuator networks," *International Journal of Control, Automation and Systems*, vol. 16, no. 3, pp. 953–960, 2018.
- [153] O. J. Faqir, Y. Nie, E. C. Kerrigan, and D. Gündüz, "Energy-efficient communication in mobile aerial relay-assisted networks using predictive control," *IFAC-PapersOnLine*, vol. 51, no. 20, pp. 197–202, 2018.
- [154] E. D'Amato, M. Mattei, and I. Notaro, "Distributed reactive model predictive control for collision avoidance of unmanned aerial vehicles in civil airspace," *Journal of Intelligent* & *Robotic Systems*, vol. 97, no. 1, pp. 185–203, 2020.
- [155] Y. Wang, X. Wang, S. Zhao, and L. Shen, "A Hierarchical Collision Avoidance Architecture for Multiple Fixed-Wing UAVs in an Integrated Airspace," *arXiv preprint arXiv:2005.14455*, 2020.
- [156] J. Dentler, M. Rosalie, G. Danoy, P. Bouvry, S. Kannan, M. A. Olivares-Mendez, and H. Voos, "Collision avoidance effects on the mobility of a uav swarm using chaotic ant colony with model predictive control," *Journal of Intelligent & Robotic Systems*, vol. 93, no. 1-2, pp. 227–243, 2019.
- [157] J. V. Saraiva, V. F. Monteiro, F. R. M. Lima, T. F. Maciel, and F. R. P. Cavalcanti, "A Qlearning based approach to spectral efficiency maximization in multiservice wireless systems," in *Proceedings of the Brazilian Telecommunications Symposium (SBrT)*, 2019.
- [158] F. Slimeni, Z. Chtourou, B. Scheers, V. Le Nir, and R. Attia, "Cooperative Q-learning based channel selection for cognitive radio networks," *Wireless Networks*, vol. 25, no. 7, pp. 4161–4171, 2019.
- [159] J. Liu, Q. Wang, C. He, K. Jaffrès-Runser, Y. Xu, Z. Li, and Y. Xu, "QMR: Q-learning based multi-objective optimization routing protocol for flying ad hoc networks," *Computer Communications*, vol. 150, pp. 304–316, 2020.
- [160] A. El-Amine, M. Iturralde, H. A. H. Hassan, and L. Nuaymi, "A distributed Q-Learning approach for adaptive sleep modes in 5G networks," in 2019 IEEE Wireless Communications and Networking Conference (WCNC), pp. 1–6, IEEE, 2019.
- [161] V. Mnih, K. Kavukcuoglu, D. Silver, A. Graves, I. Antonoglou, D. Wierstra, and M. Riedmiller, "Playing atari with deep reinforcement learning," *arXiv preprint arXiv:1312.5602*, 2013.
- [162] N. C. Luong, D. T. Hoang, S. Gong, D. Niyato, P. Wang, Y.-C. Liang, and D. I. Kim, "Applications of deep reinforcement learning in communications and networking: A survey," *IEEE Communications Surveys & Tutorials*, vol. 21, no. 4, pp. 3133–3174, 2019.
- [163] Y. Qian, J. Wu, R. Wang, F. Zhu, and W. Zhang, "Survey on reinforcement learning applications in communication networks," *Journal of Communications and Information Networks*, vol. 4, no. 2, pp. 30–39, 2019.



- [164] Z. Xiong, Y. Zhang, D. Niyato, R. Deng, P. Wang, and L. Wang, "Deep Reinforcement Learning for Mobile 5G and Beyond: Fundamentals, Applications, and Challenges," *IEEE Vehicular Technology Magazine*, vol. 14, no. 2, pp. 44–52, 2019.
- [165] Y. L. Lee and D. Qin, "A Survey on Applications of Deep Reinforcement Learning in Resource Management for 5G Heterogeneous Networks," in 2019 Asia-Pacific Signal and Information Processing Association Annual Summit and Conference (APSIPA ASC), pp. 1856–1862, IEEE, 2019.
- [166] S. S. Doddalinganavar, P. Tergundi, and R. S. Patil, "Survey on Deep Reinforcement Learning Protocol in VANET," in *2019 1st International Conference on Advances in Information Technology (ICAIT)*, pp. 81–86, IEEE, 2019.
- [167] Y. Wang, Z. Ye, P. Wan, and J. Zhao, "A survey of dynamic spectrum allocation based on reinforcement learning algorithms in cognitive radio networks," *Artificial Intelligence Review*, vol. 51, no. 3, pp. 493–506, 2019.
- [168] D. Liu, C. Sun, C. Yang, and L. Hanzo, "Optimizing wireless systems using unsupervised and reinforced-unsupervised deep learning," *IEEE Network*, 2020.
- [169] N. Naderializadeh, J. Sydir, M. Simsek, and H. Nikopour, "Resource Management in Wireless Networks via Multi-Agent Deep Reinforcement Learning," *arXiv preprint arXiv:2002.06215*, 2020.
- [170] R. Atallah, C. Assi, and M. Khabbaz, "Deep reinforcement learning-based scheduling for roadside communication networks," in 2017 15th International Symposium on Modeling and Optimization in Mobile, Ad Hoc, and Wireless Networks (WiOpt), pp. 1–8, IEEE, 2017.
- [171] S. Chinchali, P. Hu, T. Chu, M. Sharma, M. Bansal, R. Misra, M. Pavone, and S. Katti, "Cellular network traffic scheduling with deep reinforcement learning," in *Thirty-Second AAAI Conference on Artificial Intelligence*, 2018.
- [172] F. Al-Tam, N. Correia, and J. Rodriguez, "Learn to Schedule (LEASCH): A Deep reinforcement learning approach for radio resource scheduling in the 5G MAC layer," arXiv preprint arXiv:2003.11003, 2020.
- [173] X. Li, J. Fang, W. Cheng, H. Duan, Z. Chen, and H. Li, "Intelligent power control for spectrum sharing in cognitive radios: A deep reinforcement learning approach," *IEEE* access, vol. 6, pp. 25463–25473, 2018.
- [174] X. Chen, C. Wu, T. Chen, H. Zhang, Z. Liu, Y. Zhang, and M. Bennis, "Age of Information Aware Radio Resource Management in Vehicular Networks: A Proactive Deep Reinforcement learning perspective," *IEEE Transactions on Wireless Communications*, vol. 19, no. 4, pp. 2268–2281, 2020.
- [175] A. T. Z. Kasgari and W. Saad, "Model-free ultra reliable low latency communication (URLLC): A deep reinforcement learning framework," in *ICC 2019-2019 IEEE International Conference on Communications (ICC)*, pp. 1–6, IEEE, 2019.



- [176] A. T. Z. Kasgari, W. Saad, M. Mozaffari, and H. V. Poor, "Experienced deep reinforcement learning with generative adversarial networks (GANs) for model-free ultra reliable low latency communication," *arXiv preprint arXiv:1911.03264*, 2019.
- [177] X. Liao, J. Shi, Z. Li, L. Zhang, and B. Xia, "A model-driven deep reinforcement learning heuristic algorithm for resource allocation in ultra-dense cellular networks," *IEEE Transactions on Vehicular Technology*, vol. 69, no. 1, pp. 983–997, 2019.
- [178] Z. Liu, X. Chen, Y. Chen, and Z. Li, "Deep reinforcement learning based dynamic resource allocation in 5G ultra-dense networks," in 2019 IEEE International Conference on Smart Internet of Things (SmartloT), pp. 168–174, IEEE, 2019.
- [179] R. Balakrishnan, K. Sankhe, V. S. Somayazulu, R. Vannithamby, and J. Sydir, "Deep Reinforcement Learning Based Traffic-and Channel-Aware OFDMA Resource Allocation," in 2019 IEEE Global Communications Conference (GLOBECOM), pp. 1–6, IEEE, 2019.
- [180] J. Wang, J. Cao, S. Wang, Z. Yao, and W. Li, "IRDA: Incremental Reinforcement Learning for Dynamic Resource Allocation," *IEEE Transactions on Big Data*, 2020.
- [181] H.-S. Lee, J.-Y. Kim, and J.-W. Lee, "Resource allocation in wireless networks with deep reinforcement learning: A circumstance-independent approach," *IEEE Systems Journal*, vol. 14, no. 2, pp. 2589–2592, 2019.
- [182] C. Zhong, Z. Lu, M. C. Gursoy, and S. Velipasalar, "A Deep Actor-Critic Reinforcement Learning Framework for Dynamic Multichannel Access," *IEEE Transactions on Cognitive Communications and Networking*, vol. 5, no. 4, pp. 1125–1139, 2019.
- [183] Y. Zhou, F. Tang, Y. Kawamoto, and N. Kato, "Reinforcement Learning-Based Radio Resource Control in 5G Vehicular Network," *IEEE Wireless Communications Letters*, vol. 9, no. 5, pp. 611–614, 2019.
- [184] H. Ye, G. Y. Li, and B.-H. F. Juang, "Deep reinforcement learning based resource allocation for V2V communications," *IEEE Transactions on Vehicular Technology*, vol. 68, no. 4, pp. 3163–3173, 2019.
- [185] X. Zhang, M. Peng, S. Yan, and Y. Sun, "Deep-Reinforcement-Learning-Based Mode Selection and Resource Allocation for Cellular V2X Communications," *IEEE Internet* of Things Journal, vol. 7, no. 7, pp. 6380–6391, 2020.
- [186] S. Gyawali, Y. Qian, and R. Q. Hu, "Resource Allocation in Vehicular Communications Using Graph and Deep Reinforcement Learning," in *2019 IEEE Global Communications Conference (GLOBECOM)*, pp. 1–6, IEEE, 2019.
- [187] I. Althamary, C.-W. Huang, and P. Lin, "A survey on multi-agent reinforcement learning methods for vehicular networks," in 2019 15th International Wireless Communications & Mobile Computing Conference (IWCMC), pp. 1154–1159, IEEE, 2019.
- [188] J. V. Saraiva, I. M. Braga Jr, V. F. Monteiro, F. R. M. Lima, T. F. Maciel, W. C. Freitas Jr, and F. R. P. Cavalcanti, "Deep Reinforcement Learning for QoS-Constrained Resource Allocation in Multiservice Networks," *arXiv preprint arXiv:2003.02643*, 2020.



- [189] T. Zhang and S. Mao, "Smart Power Control for Quality-Driven Multi-User Video Transmissions: A Deep Reinforcement Learning Approach," *IEEE Access*, vol. 8, pp. 611– 622, 2019.
- [190] Z. Xu, Y. Wang, J. Tang, J. Wang, and M. C. Gursoy, "A deep reinforcement learning based framework for power-efficient resource allocation in cloud RANs," in 2017 IEEE International Conference on Communications (ICC), pp. 1–6, IEEE, 2017.
- [191] D. Liu, J. Zhao, C. Yang, and L. Hanzo, "Accelerating Deep Reinforcement Learning With the Aid of a Partial Model: Power-Efficient Predictive Video streaming," *arXiv* preprint arXiv:2003.09708, 2020.
- [192] D. Liu, J. Zhao, and C. Yang, "Energy-saving predictive video streaming with deep reinforcement learning," in 2019 IEEE Global Communications Conference (GLOBE-COM), pp. 1–6, IEEE, 2019.
- [193] Z. Lu and M. C. Gursoy, "Dynamic channel access and power control via deep reinforcement learning," in 2019 IEEE 90th Vehicular Technology Conference (VTC2019-Fall), pp. 1–5, IEEE, 2019.
- [194] Y. Fan, Z. Zhang, and H. Li, "Message passing based distributed learning for joint resource allocation in millimeter wave heterogeneous networks," *IEEE Transactions* on Wireless Communications, vol. 18, no. 5, pp. 2872–2885, 2019.
- [195] N. Zhao, Y.-C. Liang, D. Niyato, Y. Pei, and Y. Jiang, "Deep reinforcement learning for user association and resource allocation in heterogeneous networks," in 2018 IEEE Global Communications Conference (GLOBECOM), pp. 1–6, IEEE, 2018.
- [196] N. Zhao, Y.-C. Liang, D. Niyato, Y. Pei, M. Wu, and Y. Jiang, "Deep reinforcement learning for user association and resource allocation in heterogeneous cellular networks," *IEEE Transactions on Wireless Communications*, vol. 18, no. 11, pp. 5141– 5152, 2019.
- [197] H. Ding, F. Zhao, J. Tian, D. Li, and H. Zhang, "A deep reinforcement learning for user association and power control in heterogeneous networks," *Ad Hoc Networks*, vol. 102, p. 102069, 2020.
- [198] J. A. Ayala-Romero, J. J. Alcaraz, A. Zanella, and M. Zorzi, "Online learning for energy saving and interference coordination in hetnets," *IEEE Journal on Selected Areas in Communications*, vol. 37, no. 6, pp. 1374–1388, 2019.
- [199] L. Zhang, J. Tan, Y.-C. Liang, G. Feng, and D. Niyato, "Deep reinforcement learning for modulation and coding scheme selection in cognitive HetNets," in *ICC 2019-2019 IEEE International Conference on Communications (ICC)*, pp. 1–6, IEEE, 2019.
- [200] L. Zhang, J. Tan, Y.-C. Liang, G. Feng, and D. Niyato, "Deep reinforcement learningbased modulation and coding scheme selection in cognitive heterogeneous networks," *IEEE Transactions on Wireless Communications*, vol. 18, no. 6, pp. 3281–3294, 2019.



- [201] R. Karmakar, S. Chattopadhyay, and S. Chakraborty, "SmartLA: Reinforcement learning-based link adaptation for high throughput wireless access networks," *Computer Communications*, vol. 110, pp. 1–25, 2017.
- [202] Y. S. Nasir and D. Guo, "Multi-agent deep reinforcement learning for dynamic power allocation in wireless networks," *IEEE Journal on Selected Areas in Communications*, vol. 37, no. 10, pp. 2239–2250, 2019.
- [203] J. Ren and J. Haupt, "A communication efficient hierarchical distributed optimization algorithm for multi-agent reinforcement learning," in *Real-world Sequential Decision Making Workshop at International Conference on Machine Learning*, 2019.
- [204] J. Koo, V. B. Mendiratta, M. R. Rahman, and A. Walid, "Deep Reinforcement Learning for Network Slicing with Heterogeneous Resource Requirements and Time Varying traffic dynamics," in 2019 15th International Conference on Network and Service Management (CNSM), pp. 1–5, IEEE, 2019.
- [205] H. Wang, Y. Wu, G. Min, J. Xu, and P. Tang, "Data-driven dynamic resource scheduling for network slicing: A deep reinforcement learning approach," *Information Sciences*, vol. 498, pp. 106–116, 2019.
- [206] M. Alsenwi, N. H. Tran, M. Bennis, S. R. Pandey, A. K. Bairagi, and C. S. Hong, "Intelligent Resource Slicing for eMBB and URLLC Coexistence in 5G and Beyond: A Deep Reinforcement Learning based approach," *arXiv preprint arXiv:2003.07651*, 2020.
- [207] K. Feng, Q. Wang, X. Li, and C.-K. Wen, "Deep reinforcement learning based intelligent reflecting surface optimization for MISO communication systems," *IEEE Wireless Communications Letters*, vol. 9, no. 5, pp. 745–749, 2020.
- [208] Q. Zhang, W. Saad, and M. Bennis, "Millimeter wave communications with an intelligent reflector: Performance optimization and distributional reinforcement learning," *arXiv preprint arXiv:2002.10572*, 2020.
- [209] H. He and H. Jiang, "Deep learning based energy efficiency optimization for distributed cooperative spectrum sensing," *IEEE Wireless Communications*, vol. 26, no. 3, pp. 32– 39, 2019.
- [210] A. Zappone, M. Di Renzo, M. Debbah, T. T. Lam, and X. Qian, "Model-aided wireless artificial intelligence: Embedding expert knowledge in deep neural networks for wireless system optimization," *IEEE Vehicular Technology Magazine*, vol. 14, no. 3, pp. 60–69, 2019.
- [211] A. Zappone, M. Debbah, and Z. Altman, "Online energy-efficient power control in wireless networks by deep neural networks," in 2018 IEEE 19th International Workshop on Signal Processing Advances in Wireless Communications (SPAWC), pp. 1–5, IEEE, 2018.
- [212] A. Al-Baidhani and H. H. Fan, "Learning for detection: A deep learning wireless communication receiver over rayleigh fading channels," in *2019 International Conference on Computing, Networking and Communications (ICNC)*, pp. 6–10, IEEE, 2019.



- [213] Y. Yang, Y. Li, K. Li, S. Zhao, R. Chen, J. Wang, and S. Ci, "DECCO: Deep-learning enabled coverage and capacity optimization for massive MIMO systems," *IEEE Access*, vol. 6, pp. 23361–23371, 2018.
- [214] Z. Han, D. Niyato, W. Saad, and T. Başar, *Game Theory for Next Generation Wireless* and Communication Networks: Modeling, Analysis, and Design. Cambridge University Press, 2019.
- [215] T. Liu, J. Li, F. Shu, Y. Du, and Z. Han, "Distributed resource allocation in cachingenabled heterogeneous cellular networks based on matching theory," *CCF Transactions on Networking*, vol. 2, no. 1, pp. 57–68, 2019.
- [216] S. A. Kazmi, N. H. Tran, T. M. Ho, A. Manzoor, D. Niyato, and C. S. Hong, "Coordinated device-to-device communication with non-orthogonal multiple access in future wireless cellular networks," *IEEE Access*, vol. 6, pp. 39860–39875, 2018.
- [217] L. Zhu, J. Zhang, Z. Xiao, X. Cao, D. O. Wu, and X.-G. Xia, "Joint power control and beamforming for uplink non-orthogonal multiple access in 5G millimeter-wave communications," *IEEE Transactions on Wireless Communications*, vol. 17, no. 9, pp. 6177– 6189, 2018.
- [218] M. Sun, X. Xu, X. Tao, P. Zhang, and V. C. M. Leung, "NOMA-Based D2D-Enabled Traffic Offloading for 5G and Beyond Networks Employing Licensed and Unlicensed Access," *IEEE Transactions on Wireless Communications*, vol. 19, no. 6, pp. 4109– 4124, 2020.
- [219] J. Tang, J. Luo, M. Liu, D. K. So, E. Alsusa, G. Chen, K.-K. Wong, and J. A. Chambers, "Energy efficiency optimization for NOMA with SWIPT," *IEEE Journal of Selected Topics in Signal Processing*, vol. 13, no. 3, pp. 452–466, 2019.
- [220] M. Moltafet, N. Mokari, M. R. Javan, H. Saeedi, and H. Pishro-Nik, "A new multiple access technique for 5G: Power domain sparse code multiple access (PSMA)," *IEEE Access*, vol. 6, pp. 747–759, 2017.
- [221] F. Moradi, E. Fitzgerald, M. Pióro, and B. Landfeldt, "Flexible DRX Optimization for LTE and 5G," *IEEE Transactions on Vehicular Technology*, vol. 69, no. 1, pp. 607–621, 2019.
- [222] P. Huang and Y. Pi, "A novel MIMO channel state feedback scheme and overhead calculation," *IEEE Transactions on Communications*, vol. 66, no. 10, pp. 4550–4562, 2018.
- [223] Y. Yang, S. Dang, Y. He, and M. Guizani, "Markov decision-based pilot optimization for 5G V2X vehicular communications," *IEEE Internet of Things Journal*, vol. 6, no. 1, pp. 1090–1103, 2018.
- [224] E. Olfat and M. Bengtsson, "Joint channel and clipping level estimation for OFDM in IoT-based networks," *IEEE Transactions on Signal Processing*, vol. 65, no. 18, pp. 4902–4911, 2017.



- [225] I.-S. Comşa, S. Zhang, M. E. Aydin, P. Kuonen, Y. Lu, R. Trestian, and G. Ghinea, "Towards 5G: A reinforcement learning-based scheduling solution for data traffic management," *IEEE Transactions on Network and Service Management*, vol. 15, no. 4, pp. 1661–1675, 2018.
- [226] I.-S. Comşa, R. Trestian, G.-M. Muntean, and G. Ghinea, "5MART: A 5G sMART scheduling framework for optimizing QoS through reinforcement learning," *IEEE Transactions on Network and Service Management*, 2019.
- [227] A. A. Esswie and K. I. Pedersen, "Opportunistic spatial preemptive scheduling for URLLC and eMBB coexistence in multi-user 5G networks," *leee Access*, vol. 6, pp. 38451–38463, 2018.
- [228] A. Anand and G. de Veciana, "Resource allocation and HARQ optimization for URLLC traffic in 5G wireless networks," *IEEE Journal on Selected Areas in Communications*, vol. 36, no. 11, pp. 2411–2421, 2018.
- [229] H. Jang, J. Kim, W. Yoo, and J.-M. Chung, "URLLC Mode Optimal Resource Allocation to Support HARQ in 5G Wireless Networks," *IEEE Access*, vol. 8, pp. 126797–126804, 2020.
- [230] P. Hu and J. Zhang, "5G-Enabled Fault Detection and Diagnostics: How Do We Achieve Efficiency?," *IEEE Internet of Things Journal*, vol. 7, no. 4, pp. 3267–3281, 2020.
- [231] T. Wang, G. Zhang, X. Yang, and A. Vajdi, "Genetic algorithm for energy-efficient clustering and routing in wireless sensor networks," *Journal of Systems and Software*, vol. 146, pp. 196–214, 2018.
- [232] R. Elhabyan, W. Shi, and M. St-Hilaire, "A Pareto optimization-based approach to clustering and routing in Wireless Sensor Networks," *Journal of Network and Computer Applications*, vol. 114, pp. 57–69, 2018.
- [233] A. Moradi, "Broadcast Routing in Wireless Ad-Hoc Networks: A Particle Swarm optimization Approach," *Caspian Journal of Mathematical Sciences (CJMS)*, vol. 7, no. 1, pp. 46–67, 2018.
- [234] B. Mao, F. Tang, Z. M. Fadlullah, N. Kato, O. Akashi, T. Inoue, and K. Mizutani, "A novel non-supervised deep-learning-based network traffic control method for software defined wireless networks," *IEEE Wireless Communications*, vol. 25, no. 4, pp. 74–81, 2018.
- [235] S. Chaudhary and R. Johari, "ORuML: Optimized Routing in wireless networks using Machine Learning," *International Journal of Communication Systems*, vol. 33, no. 11, p. e4394, 2020.
- [236] M. D. F. De Grazia, D. Zucchetto, A. Testolin, A. Zanella, M. Zorzi, and M. Zorzi, "QoE multi-stage machine learning for dynamic video streaming," *IEEE Transactions* on Cognitive Communications and Networking, vol. 4, no. 1, pp. 146–161, 2017.



- [237] M. Dalla Cia, F. Mason, D. Peron, F. Chiariotti, M. Polese, T. Mahmoodi, M. Zorzi, and A. Zanella, "Using smart city data in 5G self-organizing networks," *IEEE Internet of Things Journal*, vol. 5, no. 2, pp. 645–654, 2017.
- [238] J. Wu, B. Cheng, M. Wang, and J. Chen, "Quality-aware energy optimization in wireless video communication with multipath TCP," *IEEE/ACM Transactions on Networking*, vol. 25, no. 5, pp. 2701–2718, 2017.
- [239] J. Heuschkel, E. Fleckstein, M. Ofenloch, and M. Muhlhauser, "UDP++: Cross-layer optimizable transport protocol for managed wireless networks," in 2019 IEEE Global Communications Conference (GLOBECOM), pp. 1–6, IEEE, 2019.



www.windmill-itn.eu